# **Exploring Quantum Materials with Resonant Inelastic X-Ray Scattering**

M. Mitrano<sup>®</sup>,<sup>1</sup> S. Johnston<sup>®</sup>,<sup>2,3</sup> Young-June Kim<sup>®</sup>,<sup>4</sup> and M. P. M. Dean<sup>®</sup>,<sup>\*</sup>

<sup>1</sup>Department of Physics, Harvard University, Cambridge, Massachusetts 02138, USA

<sup>2</sup>Department of Physics and Astronomy, The University of Tennessee, Knoxville, Tennessee 37966, USA

<sup>3</sup>Institute for Advanced Materials and Manufacturing, The University of Tennessee,

Knoxville, Tennessee 37996, USA

<sup>4</sup>Department of Physics, University of Toronto, Toronto, Ontario M5S 1A7, Canada

<sup>5</sup>Condensed Matter Physics and Materials Science Department, Brookhaven National Laboratory,

Upton, New York 11973, USA

(Received 11 February 2024; revised 31 July 2024; published 13 December 2024)

Understanding quantum materials—solids in which interactions among constituent electrons yield a great variety of novel emergent quantum phenomena—is a forefront challenge in modern condensed matter physics. This goal has driven the invention and refinement of several experimental methods, which can spectroscopically determine the elementary excitations and correlation functions that determine material properties. Here we focus on the future experimental and theoretical trends of resonant inelastic x-ray scattering (RIXS), which is a remarkably versatile and rapidly growing technique for probing different charge, lattice, spin, and orbital excitations in quantum materials. We provide a forward-looking introduction to RIXS and outline how this technique is poised to deepen our insight into the nature of quantum materials and of their emergent electronic phenomena.

DOI: 10.1103/PhysRevX.14.040501

Subject Areas: Condensed Matter Physics

## I. INTRODUCTION

## A. Challenges posed by quantum materials

"Quantum materials" is a relatively new classification intended to unify a diverse set of materials governed by many-body interactions and quantum mechanics manifested over large energy ranges and length scales [1–3]. While the relevant interactions in these materials are known and relatively simple, their quantum many-body nature gives rise to a host of emergent phenomena, including unconventional superconductivity [4], unconventional or strange metallicity [5], quantum spin liquids (QSLs) [6], nonequilibrium states without equilibrium analogs [7], and beyond.

Many quantum materials exhibit strong electron-electron interactions, which means that they often cannot be modeled accurately using textbook single-particle approximations. Instead, their correlated nature makes them prone to striking collective responses and extremely sensitive to external tuning parameters like applied fields, pressure, or doping. It is also common for different degrees of freedom to become entangled or intertwined in nontrivial ways,

Contact author: mdean@bnl.gov

making it difficult to identify which interaction(s) drive a particular phenomenon. While this sensitivity represents a powerful asset for technological applications, it challenges many established concepts in condensed matter physics and hinders our predictive capabilities. For example, we still lack a comprehensive framework for many of the correlated electron liquids that emerge from doping different ordered states. Solving this issue will require a deeper understanding of the elementary charge excitations that reflect a material's dynamics, and is a key step toward determining how and why unconventional superconductivity emerges from these correlated states [4]. More broadly, the rich physics of quantum materials poses fundamental questions about how to characterize and ultimately control their quantum states. For example, how do we definitively detect and quantify quasiparticle fractionalization and quantum entanglement in pursuit of realizing QSLs? How can we determine which degrees of freedom are essential to a particular state, which are helpers, and which are irrelevant?

Beyond addressing fundamental questions about the nature of condensed matter, the field of quantum materials also strives to develop completely new states and functionalities. Advances in laser technology offer routes to hybridize materials with strong light fields to form strongly correlated electron-photon systems [8]. How can we use these light-matter interactions to engineer novel transient phases? The synthesis of artificial materials and heterostructures has enabled us to observe emergent properties by

Published by the American Physical Society under the terms of the Creative Commons Attribution 4.0 International license. Further distribution of this work must maintain attribution to the author(s) and the published article's title, journal citation, and DOI.



FIG. 1. The Kramers-Heisenberg process for resonant inelastic x-ray scattering (RIXS) and the different excitations that it can probe. The RIXS process, shown in the center, involves the resonant absorption of an x-ray photon, creating an intermediate state with a core hole and a valence excitation, before the hole is filled via the emission of another x-ray photon. By measuring the energy and momentum change of the x rays, one can infer the properties of the excitations created in the material. Around the outside, we illustrate the many different types of excitation that RIXS can probe, arranged clockwise in order of increasing energy scale, as denoted by the red-to-blue circular arrow.

exploiting proximity effects between different quantum materials in reduced dimensions. The search for new functional quantum materials has been further invigorated by the discovery of magnetic van der Waals materials, which provide platforms for new types of functional magnetic excitons. These excitations offer new routes for transducing magnetic and optical information and for realizing switchable electronic devices [9,10]. The path to understanding and manipulating magnetic van der Waals materials presents new challenges involving accessing small sample volumes and buried interfaces, but this emerging area is ripe for new discoveries at the interface of applied and fundamental science.

Solving these frontier issues in quantum materials research will ultimately require us to unravel the interplay of the relevant electronic interactions along with aspects of topology, low dimensionality, and subtle lattice effects. To this end, it has been fruitful to conceptualize different quantum materials using effective models that describe the "essential" aspects of a given system, namely their ground state and low-energy elementary quasiparticle and collective excitations [11,12]. Starting from experiments that directly measure the properties of elementary collective excitations (magnons, phonons, orbitons, etc.), and in particular their momentum dependence, one can construct highly predictive effective theories with relatively few input parameters. However, the success of this approach is critically dependent on the availability of high-quality momentum-resolved experimental probes capable of accessing the elementary excitations of spin, charge, orbital, and lattice degrees of freedom. This need has motivated the development of a host of novel experimental techniques. This Perspective shines a light on one such technique, resonant inelastic x-ray scattering (RIXS) (see Fig. 1), which has experienced rapid growth as a momentum-resolved probe of quantum materials and is endowed with unique capabilities for interrogating their collective excitations. Progress in instrumentation means that we are now at a watershed period of being able to apply RIXS with time and energy resolutions that match the fundamental energy scales of many quantum materials and solve key problems in this major area of condensed matter physics.

#### **B.** Role of **RIXS** in the experimenter's toolbox

An effective spectroscopic probe of quantum materials should ideally possess several attributes. Given that these solids feature multiple charge, lattice, spin, and orbital degrees of freedom, it is advantageous to be able to address all of these selectively. Furthermore, since many elementary excitations have propagating character and are thus strongly momentum dependent, this tool should possess good momentum and energy resolution. Many questions in quantum materials hinge on measuring the properties of small samples. Therefore, it is also useful to have a probe with a suitably small focal spot size, while still being able to penetrate beneath the surface of a sample to access bulk properties. Finally, the flexibility to measure in extreme environments, such as *in situ* during applied electrical, magnetic, or strain fields, in a time-resolved modality after laser excitation, or within specific layers of a heterostructure, opens unique scientific opportunities. With some caveats, RIXS can help address all these needs.

The central panel of Fig. 1 illustrates the RIXS process. In RIXS, a monochromatic beam of x rays is directed onto a material with an energy tuned to an atomic transition between a deep core state and an unoccupied valence state. The incoming photons are absorbed and excite a core electron into an unoccupied state, before the core hole is filled via the emission of another photon. By measuring the energy and momentum change of the scattered x rays, one can extract the momentum-dependent excitation spectrum of the material. As we will explain later, the resonance process provides a key advantage of RIXS—that it couples to magnetic, orbital, lattice, and charge modes as shown in Fig. 1. It also makes RIXS element selective, which means that it can, for example, selectively study different layers in a heterostructure.

Being an x-ray photon-in, photon-out process gives RIXS several advantages. The x-ray wavelength is comparable to the interatomic spacing in materials, which enables the direct measurement of the dispersion of excitations, a capability beyond the reach of infrared [13] or Raman spectroscopy [14]. The x rays also penetrate between a few microns and about a tenth of a micron into the sample, depending on their energy, making the technique bulk sensitive. The x-ray beams can also be focused tightly into small spots, which is needed for studying small samples. Finally, RIXS also benefits from pulsed high-intensity beams produced by modern x-ray free-electron lasers (XFELs) to perform ultrafast pump-probe experiments.

#### C. RIXS cross section

Because RIXS involves a resonant scattering process, it is commonly described using the Kramers-Heisenberg (KH) formalism, which is the result of treating the photon-matter interaction using second-order perturbation theory [15]. Denoting the momentum, energy, and polarization of the incoming (outgoing) x rays as  $\hbar k$ ,  $\hbar \omega_k$ , and  $\hat{\epsilon}$ ( $\hbar k'$ ,  $\hbar \omega_{k'}$ , and  $\hat{\epsilon}'$ ), respectively, the intensity for RIXS is proportional to

$$I \propto \sum_{f} |M_{fi}|^2 \delta(E_f - E_i - \hbar \omega_q).$$
(1)

In this process we define photon energy loss  $\hbar \omega_q = \hbar(\omega_k - \omega_{k'})$  and scattering vector  $\hbar q = \hbar(k - k')$ .  $M_{fi}$  is the matrix element from the system's initial state *i*, with energy  $E_i$ , to its final state *f*, with energy  $E_f$ , via an intermediate state *n* with a core hole. In the KH approach, the matrix element is given by

$$M_{fi} = \sum_{n} \frac{\langle f | \mathcal{D}_{k',\hat{c}'}^{\dagger} | n \rangle \langle n | \mathcal{D}_{k,\hat{c}} | i \rangle}{E_n - E_i - \hbar \omega_k + \mathrm{i} \Gamma_n / 2}, \qquad (2)$$

where  $\Gamma_n/2$  is the inverse core-hole lifetime in units of energy,  $E_n$  is the energy of the intermediate state, and  $\mathcal{D}_{\mathbf{k},\hat{\epsilon}}$  $(\mathcal{D}_{\mathbf{k}',\hat{\epsilon}'}^{\dagger})$  is the absorption (emission) operator for the appropriate edge. Since the x-ray wavelength tends to be larger than the extent of the atomic orbitals, these operators are often evaluated using the dipole approximation.

While Sec. VII discusses the theoretical treatment of the RIXS cross section, Eqs. (1) and (2) already provide several insights. The scattering process involves a quantum-mechanical superposition of all possible intermediate states  $|n\rangle$  distributed throughout the lattice, which is why the collective, momentum-dependent excitations of a material can be measured. In this sense, RIXS is related to the more established technique of inelastic neutron scattering (INS), which is a powerful probe of magnetic excitations and phonons [17] (see also Box 1).

A common approach to treating experimental scattering cross sections is to map them onto multiparticle response functions (which depend only on the material) multiplied by prefactors accounting for the probe's attributes. Angleresolved photoemission (ARPES) reflects the singleparticle spectral function multiplied by a factor related to the dipole interaction of incident light with the material [24,25]. Magnetic INS, on the other hand, involves a twospin correlation function multiplied by a simple analytical prefactor that depends on the neutrons [17]. This description thus streamlines theoretical calculations and enables them to be directly compared with experimental observations. RIXS is different and, as we discuss in Sec. VII, there is no completely general way to make such a simplification for this probe. Often, elementary excitations are identifiable from basic considerations, such as the nature of the material under study or the energy scale and dispersion of the excitation itself, but this is not always the case.

The RIXS cross-section complexity, however, provides opportunities to extract additional information. The visibility of certain excitations depends on the type of corehole resonance such that specific excitations can be isolated through the judicious selection of a particular resonance energy. For example, choosing an intermediate state with a 2p core hole, which features strong spin-orbit coupling (SOC), allows the exchange of the orbital angular momentum of the photon with the spin angular momentum in the valence band to directly create spin-flip excitations [26].

# BOX 1 Comparison between RIXS and other scattering techniques

RIXS is just one member of a family of scattering techniques suited to measuring the collective excitations of quantum materials, distinct from single-particle probes such as ARPES. In some ways, its closest cousin is inelastic x-ray scattering (IXS), a nonresonant technique. Away from core-level resonances, x rays interact with the sample primarily through the Thomson cross section, where the x ray's electric field interacts with all the valence and core electrons in the material. Since most electrons are tightly bound to nuclei deep within the atoms, IXS is a highly effective probe of phonons [29]. It also couples to plasmons, but the cross section is a function of the ratio of valence to core electrons making such measurements practical only for materials made up of light elements [30]. IXS also has negligible coupling to many of the other excitations shown in Fig. 1, including magnetic excitations. The x-ray energy used in IXS can be chosen to optimize the energy resolution of the experiment, rather than being fixed by the core-hole resonance, and IXS setups can achieve sub-meV energy resolution using spectrometers based on backscattering crystal optics [29].

Electron energy loss spectroscopy (EELS) is another analogous technique involving the inelastic scattering of electrons rather than photons. Because electrons are charged particles, they interact very strongly with the electronic clouds of a solid and are naturally sensitive to a larger variety of valence excitations [31,32]. EELS experiments are performed in both transmission and reflection geometries. Transmission EELS measurements have a rather simple cross section [11,33] and rely on keV-energy electrons to penetrate into the bulk of nanometer-thick samples. Reflection EELS (REELS) experiments typically involve lower energy ( $\sim$ 10–100 eV) electrons, which scatter off the surface of solids. Owing to the lower energy, REELS experiments can straightforwardly achieve meV energy resolution. However, the interaction with the surface can complicate the interpretation of the cross section of bulk quantum materials and limit access to out-of-plane momenta [32,34].

Another important technique is inelastic neutron scattering (INS). Neutrons scatter from both the atomic nuclei and valence electron magnetic moments, which interact with the neutron's magnetic moment. As such, INS is well suited to measuring phonons and magnetic excitations. The weak interactions between thermal neutrons and solids allows one to connect the signal to two-spin correlation functions [16], which facilitates the straightforward quantitative interpretation of data. However, the same weak interactions also mean that large samples are needed to obtain adequate signals. INS is especially well suited to studying very low energy excitations and can be routinely applied with sub-meV or even  $\mu$ eV resolution.

When comparing RIXS with these other scattering methods, it is important to note that the achievable performance in RIXS experiments is strongly dependent on the incident x-ray energy, including whether the energy falls within the soft, tender, or hard x-ray regimes. Most high-resolution RIXS experiments on quantum materials today are performed with energy resolution of 20–40 meV, which is well matched with charge or orbital excitations and magnetic excitations in materials with large exchange energies. The x-ray energy also determines the accessible momentum range. Typically, tender or hard x-ray RIXS can cover the whole (and multiple) Brillouin zone (BZ) and soft x-ray RIXS often only provides partial coverage.

These techniques also offer different possibilities for extreme sample environments. The penetrating power of neutrons makes INS relatively easy to implement at high magnetic field and dilution fridge temperatures, while small beam size makes hard x-ray RIXS compatible with studying materials under high pressure using a diamond anvil cell. The vacuum requirements, the limited availability of window materials, and the proximity of complex goniometers and/or spectrometer components makes extreme sample environments more difficult (but far from impossible) to implement for soft x-ray RIXS. RIXS is typically performed in the 10–300 K range and under either zero or modest magnetic fields of a fraction of a Tesla. Improving these aspects of the sample environment is particularly important for the study of QSLs. EELS is usually performed in vacuum and over a similar temperature range, but cannot be combined with high pressures (due to the electron penetration depth) and applied magnetic fields (since these would dramatically deflect the electrons).

At a 1*s* core-hole resonance, on the other hand, single spinflip excitations are forbidden [27]. Specific choices of the incoming and scattered x-ray polarization can selectively excite spin or orbital excitations according to the x-ray absorption and emission operators [28]. The choice of the resonance can also emphasize either direct "operator" RIXS, where the excitations arise from the dipole operators, or indirect "shakeup" RIXS, where excitations are created largely by interactions between the core hole and the valence electrons. Most excitingly, this richness allows RIXS to access excitations that cannot or have not been measured in any other way. Examples of these include dispersive orbital excitations called "orbitons" [29], quintuple magnons [30,31], excitations involving coupled charge-density waves (CDWs) and phonons [32], and fractional excitations in Kitaev materials [33].

## **D.** Outline

The goal of this Perspective is to shine a spotlight on key topics for future RIXS studies of quantum materials. In light of this focus, we refer any reader interested in applications to liquids, gases, or solid-state chemistry to recent reviews on these topics [34,35]. Section II also provides introductory historical context on the development of RIXS as a probe of quantum materials and Box. 1 provides some context comparing RIXS to other techniques. A reader who is already familiar with this material can safely skip these sections.

The remainder of this Perspective focuses on a selection of topics that we consider to be particularly important for the field of quantum materials. Section III explores how RIXS measurements of charge excitations will contribute to the paradigmatic challenge of understanding the anomalous properties of strongly correlated metals, including so-called "strange metals," states characterized by poorly understood electrical conduction and symmetry-breaking properties. Section IV addresses QSL materials and outlines RIXS's prospects for detecting their novel excitations, which will provide new insights into their unique properties like carrier fractionalization and many-body entanglement. The advent of XFELs has opened remarkable new prospects for understanding and controlling dynamical phenomena arising from light-matter interactions. In light of this, Sec. V examines future opportunities with time-resolved RIXS (trRIXS), which is uniquely positioned to study the collective excitations of these driven systems and prove the existence of novel types of nonequilibrium states. Section VI examines what insights RIXS can bring to the study of functional quantum materials. Here, we place a particular focus on novel excitons (collective excitations of bound electron-hole pairs) and their dynamics, which offer potential for fundamentally new forms of magneto-optical coupling. Finally, Sec. VII covers key concepts and challenges in the theory of RIXS, which will be crucial to advancing the field.

## **II. A VERY BRIEF HISTORY OF RIXS**

As with any subject, the history of RIXS is helpful in understanding future trends in this area of research. RIXS is a challenging experimental technique at the forefront of x-ray science. Being a second-order process, it has a weak cross section, necessitating a bright x-ray source and an efficient detection scheme. The energy scale of x-ray photons is also far larger than the typical excitation energies in quantum materials, which means that high resolving powers (defined as the incident photon energy divided by the energy resolution) are required. For these reasons, the development of RIXS is in large part dictated by the pace of advances in x-ray sources and spectrometers and can be thought of as a synergistic evolution of technological and scientific breakthroughs. In judging this progress, it is useful to consider that a typical order-of-magnitude temperature scale for phase transitions in many quantum materials is  $T \sim 100$  K, which corresponds to an energy scale of  $k_{\rm B}T \sim 10$  meV and a timescale of  $\hbar/(k_{\rm B}T) \sim 100$  fs (where  $k_{\rm B}$  is Boltzmann's constant). A key reason why RIXS is increasingly influential in the field of quantum materials is that its resolution is now approaching this important threshold. However, energy resolution remains a limiting factor in many types of experiments.

The way that RIXS instrumentation energy resolves x rays falls into two general types, depending on the energy of the x-ray resonance they target. This means that RIXS experimentalists must carefully consider the resonant energy of the target core-hole transition. For so-called "hard" x-ray energies above about 5 keV, x-ray wavelengths ( $\leq 3$  Å) are comparable to the lattice spacing of nearly perfect crystals such as silicon, and spectrometers in this energy range are typically based on crystal analyzers. Below about 2 keV, x-ray wavelengths are too long for this to work effectively, and soft x-ray spectrometers are consequently based around gratings [36–39]. Efforts to perform experiments in the more difficult energy range of about 2–5 keV, in what is sometimes called the tender x-ray regime, are opening compelling new research opportunities [40–42].

Some of the first RIXS research dates to the 1970s, when Eisenberger, Platzman, and collaborators demonstrated the resonant enhancement of inelastic x-ray scattering from copper metal [43,44]. However, the energy resolution at the time (0.8 eV) made it quite difficult to infer new details about copper's electronic properties. In the mid-1990s, RIXS started becoming an incisive probe of quantum materials, fueled by the marriage of improved insertion device x-ray sources and dedicated spectrometers alongside the realization that it can directly measure the energy of electronic transitions. Initial studies focused on band insulators and semiconductors [19,45–47]. Not long after, the exploration of materials such as NiO [48–50] and CeO<sub>2</sub> [51] initiated a new domain of RIXS research focused on materials with strong electronic correlations.

Buoyed by interest in strongly correlated quantum materials and cuprate high-temperature superconductors in particular [52], as well as developments in theory, RIXS has evolved from a niche to a flourishing field. A series of works in the late 1990s and early 2000s established its unique capabilities to resolve the momentum dependence of insulating "Mott" gap excitons [53-56] and the crystal field of cuprates [57]. Driven by a 2-orders-of-magnitude improvement in both resolution and count rates [16,36-39,58], the next decade saw breakthrough experimental demonstrations that RIXS can measure magnetic excitations [26,27,59,60]. These investigations focused first on undoped and underdoped cuprates [26,61] but quickly spread to optimally doped, overdoped, and electron-doped compositions [62–66], as well as to other magnetic materials [67-69]. Here, RIXS was key in clarifying the nature of



FIG. 2. The charge excitation spectrum of correlated metals holds the key to understanding their anomalous electronic transport and symmetry-breaking properties. (a) Electronic phase diagram of the cuprates showing the emergence of the unconventional or strange metal phenomenology from neighboring CDW, superconducting, and magnetic phases. (b) Conceptual picture of the charge excitations of cuprates, which reflect the properties of these unusual metallic states. At low momentum, cuprates feature plasmons, but at higher momentum, these decay to form a continuum in poorly understood ways. Cuprates also universally exhibit CDW symmetry breaking and correlations. (c) Plasmon dispersion and decay in  $Nd_{2-x}Ce_xCuO_4$  [99]. (d) Low-energy CDW excitations in  $La_{2-x}Ba_xCuO_4$  above the CDW transition [80]. Understanding these excitations can be key to comprehending unconventional metallicity.

magnetic correlations and their relationship with hightemperature superconductivity [52]. A similar explosion of activity occurred around iridates [70–75]. RIXS was key to establishing how strong SOC in these materials can realize novel types of correlated insulating and magnetically frustrated states [76,77]. Today, measuring spin excitations and magnetic interactions remains one of RIXS's main applications within quantum materials research.

Major results that go beyond magnetic excitations included the identification of dispersive orbital excitations or orbitons [29], which arise from a special quantum effect of "spin-orbital separation" in one dimension. RIXS was also central to establishing the ubiquitous presence of CDWs in cuprates [32,78–85], supporting the contention that this symmetry breaking is a fundamental aspect of the physics of these materials. Demonstrations that RIXS can probe electron-phonon coupling (EPC) [86–88], spin excitations in atomically thin layers within heterostructures [89] or on substrates [90], and ultrafast transient states [91,92] have further established some of the unique capabilities and potential of the technique.

### **III. STRONGLY CORRELATED METALS**

Anomalous metallic behavior beyond the Fermi liquid paradigm is ubiquitous in quantum materials [93]. For example, many compounds exhibit unconventional electron-electron scattering that causes their resistivity to scale linearly with temperature over a very large range, unlike the  $T^2$  dependence predicted by Fermi liquid theory. This phenomenology is often called strange metallicity [5,94]. Some of these materials even exhibit resistivity exceeding the Mott-Ioffe-Regel limit, where the mean free path of electrons is shorter than the interatomic spacing, which is often termed "bad" metal behavior [95]).

The archetype for this physics is the cuprates, where strange metal behavior emerges in the vicinity of other types of novel electronic behavior, including CDW symmetry breaking and unconventional superconductivity [see Fig. 2(a)]. Moreover, these types of phenomena also appear in a host of other strongly correlated materials, including ruthenates [96], iron pnictides [97], heavy fermion materials [93], and even magic-angle twisted bilayer graphene [98]. Given the universality of this unconventional metal phenomenology, many researchers believe understanding it will require fundamentally new physical concepts or mathematical machinery [4]. To date, RIXS has played an essential role in deepening our understanding of correlated quantum materials, with the cuprates often acting as a touchstone in this endeavor. Looking toward the future, it will continue to provide momentum- and energy-resolved measurements of the collective charge excitations across many of these systems, which will be crucial for formulating and ultimately validating new theoretical frameworks for correlated metals.

#### A. Charge excitations

At a fundamental level, unconventional metal phenomenology suggests that the collective charge excitations in cuprates are *radically different* from those predicted in the standard Fermi liquid theory [5]. Understanding these excitations may hold the key to solving the strange metal problem [100] and in formulating theories to explain other novel phenomena observed in unconventional metals [101].

In the long-wavelength  $(q \rightarrow 0)$  limit, metals typically exhibit well-defined electronic excitations called plasmons. These are collective oscillations in the electron density that primarily reflect the compressibility of the electron system. Within a Fermi liquid picture, the plasmon is expected to disperse as a function of momentum, before it intersects with a continuum of particle-hole excitations, whereupon it decays. Optics experiments have shown that the low-q charge excitation spectra of cuprates indeed feature plasmons [102,103]. RIXS experiments have recently measured their low-q dispersion in hole- and electron-doped cuprates as a function of both in-plane and out-of-plane q, as plotted in Figs. 2(b) and 2(c) [99,104–108]. These plasmons, however, are broad and decay anomalously at relatively low q before they intersect the particle-hole continuum [105,109,110]. The reasons for this remain poorly understood and are not accurately predicted by the established models.

Determining the detailed form of the high-momentum charge continuum in the cuprates and how it relates to different theories for unconventional metals and the strange metal phenomenology is an area where RIXS is likely to make decisive contributions. An important challenge is the need to isolate the charge response from the spin response and the lattice, which may be addressed by approaches such as x-ray polarimetry or some of the theoretical advances that we anticipate in Sec. VII. EELS is also likely to bring an important complementary perspective to these questions, as it has better energy resolution compared to RIXS and a simpler cross section [23]. However, its surface sensitivity presents additional challenges and limits its access to the out-of-plane plasmon dispersion.

An especially topical route to understanding cuprates is to contrast them with non-copper-containing materials with closely related properties, which will help identify the key ingredients behind their novel phenomenology. The past few years have seen the discovery of superconductivity and unconventional metallicity in nickelates, including those formed from square planar nickelate oxide layers with a low electronic valence of  $d^{9-\delta}$  [111–113] and also those built from bilayers of nickel-oxygen octahedra [114,115]. Synthesizing these materials as large single crystals and preparing pristine surfaces has proven quite challenging, so RIXS has had an outsized role in characterizing their electronic properties including the nature of the electronic correlations and the role of oxygen orbitals [116–119]. Although theoretical predictions for plasmons in these nickelates are available [120], to date there are no published experimental measurements. Looking toward the future, fully characterizing the collective excitation spectrum of nickelates and comparing their phenomenology with that of cuprates will constitute an important contribution toward the solution of the strange metal problem. Another element of this endeavor will be clarifying the currently controversial presence and nature of electronic symmetry breaking in the same materials, which is discussed next [121–125].

### **B.** Electronic symmetry breaking

Several decades of study of cuprates has established that charge and spin symmetry breaking are ubiquitous features of these materials [126,127]. Different types of charge and spin symmetry breaking are also common features of other strongly correlated metals such as nickelates, pnictides, and heavy fermion materials [128]. Modern numerical methods predict that charge order is also a feature of simple effective models such as the Hubbard model, and support the idea that it can arise from strong electronic correlations [129-132]. From a physical point of view, doped carriers can break fewer magnetic bonds if they cluster together, leading to an effective attraction between holes. This effective interaction is countered by Coulomb repulsion and hopping, which both tend to distribute carriers evenly through the crystal to minimize the total energy. The fact that these competing interactions act on different length scales implies that the overall minimum energy solution of this complicated situation may involve a modulated clustering: a CDW. Here, the term CDW correlation refers to the general CDW phenomenology that includes static long- and short-range order, as well as dynamic CDW fluctuations.

CDW correlations in cuprates interact strongly with other electronic phases. Although static charge and spin order appear to suppress superconductivity, some researchers have suggested that dynamic fluctuations may act to promote it [133,134]. Scattering from CDW fluctuations may have a role in the anomalous resistivity of strange metals [135]. RIXS has exceptional sensitivity to weak CDW correlations, as it benefits from the resonant enhancement of valence electron modulations, while using a spectrometer to reject the strong x-ray fluorescence that limits the sensitivity of traditional resonant soft x-ray scattering experiments [78,136]. Recently, RIXS has shown that diffuse CDW correlations extend to much higher temperatures and dopings than previously thought [80-83,85,137,138]. Figure 2(d) plots a RIXS measurement of La<sub>2-x</sub>Ba<sub>x</sub>CuO<sub>4</sub> showing a broad feature centered around H = -0.27 r.l.u. and modification of the continuum at finite energies [80].

While the widespread presence of CDW correlations is well established, the nature and energy of the low-energy excitations associated with the CDW q vectors are not. Determining the properties of possible CDW excitations should help definitively unravel the interactions underlying CDW formation. RIXS holds promise to disentangle these effects by examining the interactions between charge and spin [80] or detailed studies of phonons as discussed in Sec. VII D [32,139,140]. This may help clarify how important structural distortions are in the formation of CDWs. Studies under applied strain can also be very



FIG. 3. RIXS can access novel types of higher-order correlations functions that can help us identify QSLs. (a) 1D Heisenberg spin chain. The AFM Heisenberg interaction J > 0 acts equally on all three spin components. (b) Schematic diagram showing the phase space of fractionalized magnetic excitations in a spin chain. The darker shaded region is the two-spinon continuum, and all shaded regions (both light and dark blue) represent the four-spinon continuum [154]. (c) Kitaev model where spins connected by an *x* bond interact via their *x* components and equivalent for the *y* and *z* bonds. (d) Experimental Ir  $L_3$ -edge RIXS data of the magnetic excitations along the chain direction in Ba<sub>4</sub>Ir<sub>3</sub>O<sub>10</sub> from Ref. [155]. The dotted lines are the two-spinon continuum boundary shown in (b). (e) Experimental RIXS spectra of a spin chain compound Sr<sub>2</sub>CuO<sub>3</sub> obtained at the oxygen *K* edge [156]. The dotted and dashed lines indicate the boundaries of the two- and four-spinon continua, respectively, as shown in (b). (f) Theoretical RIXS intensity of the Kitaev honeycomb model in the spin-conserving channel showing a strongly structured spectrum consistent with spin fractionalization [33]. The symmetry notation for the BZ is shown above the plot.

insightful, especially if a material is on the verge of long-range CDW order [141,142].

Another major question is whether the cuprate phase diagram hosts a possible CDW quantum critical point or CDW quantum criticality in general. Near a quantum critical point, we expect a divergence of the fluctuations associated with the ordering, which can mediate unconventional forms of superconductivity [143]. Thermodynamic and transport probes indeed provide evidence of quantum criticality, but the location of these effects in the doping phase diagram suggests that quantum criticality might be more related to the pseudogap than to a CDW [144,145]. RIXS studies of how CDW excitations scale with temperature nonetheless suggest that quantum critical effects are at play in the cuprates [140,146,147]. Determining the role of quantum criticality in the cuprates will require the identification of universal effects that are present in all superconducting cuprate families. Given that the characteristic temperature of these effects is about 100 K, it will be important to obtain a better energy resolution in the few meV energy range rather than in the few tens of meV range. Such resolutions are being targeted in RIXS instruments that are currently under construction [148].

### **IV. QUANTUM SPIN LIQUIDS**

QSLs are a fundamental topic in magnetism. These are magnetic materials in which quantum fluctuations suppress the tendency to form classical long-range magnetic order, even in the limit of zero temperature [6,149,150]. To understand QSLs it is useful to consider a 1D spin-1/2Heisenberg antiferromagnet model, as shown in Fig. 3(a). The model's Hamiltonian is well understood theoretically and captures the essential physics of several real materials like Sr<sub>2</sub>CuO<sub>3</sub> [151–153]. Instead of a Néel ordering and magnon excitations, as expected from a classical picture, the system's ground state is an entangled quantum manybody state with fractional quasiparticle excitations called spinons. Since these fractionalized quasiparticles carry spin 1/2, the magnetic excitation spectrum of a material hosting spinons involves exciting even numbers of spinons, creating broad continua, as plotted in Fig. 3(b).

The key goal of QSL research is to definitively realize quantum entangled states with fractional quasiparticles in two-dimensional (2D) or three-dimensional (3D) materials. A well-known way to do this is through geometric frustration, which is found when no ordered state can simultaneously satisfy all the interactions between different spin pairs. Another type of frustration that has drawn much attention recently is exchange frustration caused by bonddirectional interactions called Kitaev interactions. Kitaev's honeycomb model shown in Fig. 3(c) is exactly solvable and features a QSL ground state with fractional excitations called Majorana fermions, a potentially revolutionary component for topological quantum computation [157,158]. Several iridates with honeycomb or honeycomblike lattices, as well as  $\alpha$ -RuCl<sub>3</sub>, have been suggested as candidates for Kitaev QSLs [159-163]. Many INS and RIXS experiments have been performed on these materials, most of which are consistent with what is expected for a QSL [73,75,164–168]. The key open challenge is to *definitively* identify QSLs. This can be difficult because these states tend to exhibit broad and somewhat featureless continua. which, in isolation, can be hard to distinguish from classical spin glasses or disordered states [169-171]. In the following, we outline two ways to address this using RIXS: The first is using special features of the RIXS cross section to identify spectroscopic signatures of electron fractionalization, and the second is to combine RIXS with concepts from quantum information to directly prove the presence of quantum entanglement.

## A. Spectroscopic signatures of QSLs

While INS is an indispensable tool for studying quantum magnetism, RIXS can provide additional information about a material's magnetic excitations. This capability arises from the RIXS cross section, which allows for excitation pathways beyond the conventional dipolar spin flips generated by INS. The cases below illustrate how the richness of the RIXS cross section will offer routes to more definitively probe QSLs.

As a first example, it is useful to compare the RIXS spectra of spinon-hosting materials taken at the L and K edges shown in Figs. 3(d) and 3(e), respectively. At the L edge, the spectrum is dominated by a two-spinon continuum yielding similar information to that obtained by INS. The K edge, on the other hand, features a strong indirect RIXS process, which produces four-spinon excitations that are well separated from the two-spinon continuum and that are not observed using INS. These higher-order correlations have a distinct structure that can be exploited to better identify QSLs. Proof of principle of this approach has been achieved by studying 1D chains [156,172], but to date, this has not been applied to 2D or 3D QSL candidates. In the same vein, the quadrupole transition of RIXS, which gives rise to  $\Delta S_{z} = 2$  excitations via double spin flips in the same atom, offers additional routes to more deeply interrogate QSL candidates with spin > 1/2. Such  $\Delta S_z = 2$  type excitations were theoretically proposed and experimentally observed in quantum magnets [16,173–175]. Recent studies reported that it is even possible to create multiple spin flips corresponding to  $\Delta S_7 = 3, 4, 5$  [30,31].

Another important distinction between INS and RIXS is that the latter can couple to an electron's charge and orbital degrees of freedom, in addition to its spin degree of freedom. This becomes particularly important for probing fractional particles such as Majorana fermions. To identify fractionalized excitations uniquely, additional information encoded in the incoming and outgoing photon polarization (a so-called polarimetry analysis) of the RIXS spectrum is valuable [33,176-182]. Theoretical predictions of this effect are shown in Fig. 3(f), which plots RIXS calculations for a Kitaev spin liquid in the "spin-conserving" polarization channel [33]. Strongly momentum-dependent spectral structure is seen coming from fractionalized quasiparitcles. This is distinct from the featureless spectrum in the "nonspin-conserving" channel, corresponding to the usual spin structure factor probed by INS. The power of polarimetry analysis was demonstrated in a study of a two-dimensional copper oxide compound, where the emergence of spinons from the decay of high-energy magnons was identified by analyzing the outgoing photon polarization [183]. One challenge in this area is that polarimetry setups have limited efficiency, so it can be difficult to perform the polarization analysis while maintaining the high-energy resolution and signal strength required to observe the desired features [184]. Advances in x-ray multilayer optics and the inclusion of collimating mirrors in modern spectrometers are expected to improve the efficiency of polarimetry measurements in the coming years [37,185]. Another challenge is that the energy resolution of RIXS is currently significantly inferior to INS (see Box 1), so these studies are likely to focus on transition oxide materials due to their comparatively large energy scales.

#### **B.** Witnessing entanglement

A major opportunity for future research lies in the marriage of scattering probes and quantum information methods. The spectral response of materials in the thermodynamic limit can be rigorously connected to the expectation values of operators called "entanglement witnesses," which certify the presence of quantum entanglement and thus have the potential to more directly and definitively test whether materials are QSLs [186-191]. Recent INS studies have demonstrated the power of this method in lowdimensional spin systems [192-196]. These studies are based on a quantity called the quantum Fisher information (QFI), which can be obtained from a sum-rule integral of the dynamical susceptibility, and which certifies multipartite entanglement if its value exceeds classical expectations. Such an approach has recently led to the identification of delafossite KYbSe<sub>2</sub> as a proximate gapped  $\mathbb{Z}_2$  QSL [195].

RIXS offers the exciting possibility of extending this type of quantum metrology to small samples and to nonequilibrium states (see Sec. V) beyond the reach of INS. A key requirement to extend the established QFI methodology to RIXS is the need to extract the magnetic dynamical susceptibility in absolute units from RIXS. This will require the identification of a standard reference cross section, such as the *dd* excitation in cuprates, combined with theory to determine the appropriate normalization factor to map the RIXS operator to the spin operator [197]. An alternative to converting RIXS to dynamical susceptibility would be to reformulate the QFI method in terms of the RIXS operator. This approach, however, requires addressing the fact that the RIXS operator in Eq. (2) is non-Hermitian [198]. Such a theory would be very impactful by opening routes to using RIXS to measure the entanglement of other degrees of freedom such as orbital entanglement.

## C. Tunable QSLs

A special area in QSL research, and quantum materials more generally, is the study of quantum critical phenomena that occur when a magnetic phase transition takes place at absolute zero [93]. A suite of novel approaches now enables the controlled modification of QSLs and access to such critical points; however, most require the use of small samples, which points to a special role for probing these systems with RIXS. An example of this would be few-layer exfoliated van der Waals materials. These may have fundamentally different microscopic properties from those in the bulk, as found in a recent study of monolayer  $\alpha$ -RuCl<sub>3</sub> [199]. An especially intriguing possibility, in the same spirit, is to control the material dimensionality via thin-film deposition techniques. This includes growing layered perovskite phases in the in-plane direction to generate 1D phases out of materials that form 2D phases in the bulk [200] or growing perovskite phases in the [111] direction to create 2D hexagonal motifs [201].

RIXS's access to small samples also facilitates studying systems subject to external tuning parameters like applied strain. Such effects have already been studied theoretically with the proposal of accessing QSLs in spin ice materials [202]. Thin film or exfoliated geometries are also suitable for producing devices and applying gating voltages to tune the required interactions and realize QSL phases.

#### V. NONEQUILIBRIUM PHASES

Strong optical fields enable new protocols for controlling quantum materials and observing remarkable cooperative responses. Recent years have seen major successes in detecting emergent phenomena such as Floquet-engineered electronic band structures [203–206] and new emergent magnetic [207–209], ferroelectric [210–212], topological [213], and superconductinglike phases [214–216]. As we argue below, many aspects of these novel states are very difficult to probe with established techniques and are consequently not well understood. The emerging technique of trRIXS has particular strengths for probing transient phases and is poised to both improve our understanding of these dynamic states of matter and to discover new phenomena. These include strongly

correlated Floquet phases such as Floquet quantum magnets with dynamically renormalized exchange interactions, broader forms of nonequilibrium condensation including  $\eta$  pairing, and electronic metastability.

### A. The dawn of time-resolved RIXS

Many photoinduced phenomena occur only on fast timescales of a few picoseconds or less, which tightly restricts how they can be measured. Optical spectroscopy is naturally well suited to interrogate dynamics at femtosecond (or even attosecond) timescales and is a very popular ultrafast technique [7,217,218]. There are, however, states that necessarily require accessing momentumdependent information, including antiferro-type ordered or gapless states. Experimental tools capable of providing momentum resolution (see Box 1) are much more limited in this context, as many probes are intrinsically challenging to translate to fast timescales. The velocity ( $\sim 10^3 \text{ ms}^{-1}$ ) and penetration depth ( $\sim 10^{-2}$  m) of thermal neutrons imply that a typical neutron scattering experiment will probe a material over a timescale of more than a microsecond, well beyond the lifetime of many light-induced states. Electrons represent a more appropriate tool for studying transient states, owing to their much shorter interaction length scale and the possibility of emitting them in short pulses. Ultrafast electron diffraction is an incisive probe of the transient crystal structure [219,220], while the photon-in, electron-out technique of time-resolved ARPES is a powerful probe of photoexcited band structures [221,222]. However, such experiments must be carefully designed to avoid so-called "space charge" effects, where the duration of the probe pulse gradually increases due to the mutual Coulomb repulsion between electrons in the bunch.

Being a photon-in, photon-out technique, RIXS is well suited to probe matter at ultrafast timescales and in rather unique ways [see Fig. 4(a)]. The wavelength of x-ray photons is short enough to track light-induced dynamics across a substantial fraction (if not the entirety) of the BZ, well beyond the zero-momentum limit of ultrafast optical spectroscopy [7,217,218]. Furthermore, owing to its rich cross section, trRIXS is sensitive to the collective electronic excitation spectrum of quantum materials in the spin, charge, and orbital degrees of freedom. These electronic excitations with bosonic character are mostly invisible to time-resolved ARPES and to ultrafast x-ray and electron diffraction. Hence, trRIXS fills a long-standing gap in ultrafast experiments and constitutes the spectroscopy of choice for probing momentum-dependent, low-energy valence excitations in driven quantum materials.

The field of trRIXS is currently at a watershed moment. These experiments require photon fluxes similar to or higher than those provided by synchrotrons, but in pulses of short duration and high intensity, which are typically generated by XFEL sources [227–229]. The first experiments tended



FIG. 4. Time-resolved RIXS will enable the discovery of transient phases without equilibrium analogs. (a) Sketch of a trRIXS experiment in which a material is perturbed by an optical pump and probed with short x-ray pulses [223]. (b) Ir *L*-edge trRIXS spectra of pseudospin excitations of Sr<sub>2</sub>IrO<sub>4</sub> after excitation with near-infrared photons [91]. (c) Cu  $L_3$ -edge trRIXS spectra of photoexcited La<sub>2-x</sub>Ba<sub>x</sub>CuO<sub>4</sub> showing prompt melting of the charge order (CO) quasielastic peak [92]. (d) Ultrafast pump fields can lift (lower) in energy the intermediate doubly occupied state of the exchange process by an amount proportional to the number of virtually absorbed (emitted) pump photons and alter the spin wave dispersion across the entire BZ [224]. *U* is the on-site Coulomb repulsion,  $t_{eff}$  the effective hopping amplitude,  $\Omega$  the pump photon energy, and *m* the Floquet index [223,224]. (e) Photoexcited low-dimensional Mott insulators are predicted to exhibit an emergent condensate called  $\eta$  pairing, i.e., a staggered superconducting phase with characteristic charge [ $N(q, \omega)$ ], spin [ $S(q, \omega)$ ], and superconducting [SC( $q, \omega$ )] response functions [225]. (f) Photodoped Mott insulators and quantum magnets can evade thermalization to a hot metallic state and give rise to nonthermal states (prethermal and metastable) if their dynamics are constrained by approximate conservation laws or symmetries [226].

to require implementing spectrometers at XFELs specially for one-off pioneering studies. Several experiments of this type have laid the foundation for the field. As shown in Fig. 4(b), antiferromagnetic (AFM)  $Sr_{n+1}Ir_nO_{3n+1}$  iridates [91,230,231] (at the Ir L edge) and CuGeO<sub>3</sub> spin chains [232] (at the O K edge) have provided the first snapshots of perturbed magnetic fluctuations at finite momentum and demonstrate that zero-momentum optical excitations reshape the spin fluctuation spectrum throughout the entire BZ. In the cuprate  $La_{2-x}Ba_xCuO_4$ , trRIXS showed that the appearance of light-enhanced superconductivity is accompanied by a prompt suppression and sudden sliding of the charge order peak [see Fig. 4(c)], which recovers through diffusive dynamics involving the annihilation of topological defects [92,233]. In the Mott-Hubbard system  $V_2O_3$ , a trRIXS study of the *dd* excitations indicated the possible presence of a nonthermal state, where the excitation of

electrons into the conduction band is accompanied by a modification of the V atom dimerization [234].

While promising, these first proof-of-principle experiments explored only a tiny fraction of the phase space opened by trRIXS. Nevertheless, they motivated investments in dedicated trRIXS facilities. The next generation of trRIXS experiments will benefit from technical advances, thanks to the simultaneous enhancement of spectral brightness through increased repetition rates [e.g., at the Linac Coherent Light Source (LCLS)-II and the European XFEL (EXFEL)] and the deployment of spectrometers with higher resolving power (of order 10 000) at LCLS-II, EXFEL, and SwissFEL. With these improvements in energy resolution and counting statistics, combined with more sophisticated and targeted photoexcitation schemes, trRIXS will be uniquely positioned to observe low-energy features of driven quantum phases at finite momentum and to gain qualitatively new insights into the microscopic physics of light-induced phenomena. The success of these experiments will depend on the development of accurate pumpprobe synchronization schemes and detectors operating close to the XFEL repetition rate to resolve coherent dynamics in the time-dependent RIXS spectra. Further, it will be crucial to devise efficient energy dissipation schemes to mitigate average heating effects due to the increased repetition rate and tighter beam focusing.

#### B. A unique view of emergent transient states

A priority research direction for trRIXS is nonequilibrium superconductivity. In recent years, intense near-infrared and midinfrared fields have led to the observation of transient states featuring superconductinglike optical and magnetic properties [214–216,235,236]. These emergent electronic states defy the conventional expectation that external perturbations destroy quantum coherence by introducing heating and often exhibit novel symmetries [237] and energy scales [216,238]. The microscopic physics of these processes is still debated, with theoretical interpretations ranging from the melting of competing orders [239], to the amplification of the superconducting instability [240], the cooling of phase fluctuations and low-energy quasiparticles [241,242], or light-induced pairing [243].

In copper oxides, light-induced superconductivity appears upon excitation with pulses tuned to the in-plane, bond-stretching [214], and apical oxygen phonons [215,244,245] or optical transitions across the chargetransfer gap [246]. Magnetic superexchange and strong oxygen copper hybridization are vital ingredients for many theories of the equilibrium superconducting state of these materials, so they represent key elements for a microscopic theory of nonequilibrium superconductivity [247]. trRIXS can directly measure the magnetic superexchange by mapping the spin fluctuations spectrum at the Cu L edge, while the charge-transfer energy is encoded in O K-edge spectra [119]. Both quantities are sensitive to the hopping matrix elements, which can be modulated either by atomic displacements due to nonlinear phonon coupling [248-250], hybridization with strong electric fields [224] [see Fig. 4(d)], or through a renormalization of the effective Coulomb repulsion [251]. Future trRIXS measurements of near- and mid-infrared-driven cuprates such as  $YBa_2Cu_3O_{6+\delta}$ , supported by the theory described in Sec. VIIE, will crucially contribute to the microscopic understanding of this intriguing phenomenon. trRIXS will also shed light on similarities and differences between light-induced superconductivity in the copper oxides and in FeSe<sub>0.5</sub> Te<sub>0.5</sub> [252], where Fe  $L_3$ -edge experiments [90] could clarify the interplay between spin fluctuations and lattice degrees of freedom in the photoexcited state [253].

By further leveraging its spin sensitivity, trRIXS will directly access light-engineered magnetic interactions in a wider variety of quantum materials and dramatically expand the field of ultrafast magnetism [7,223,230]. Nonresonant pump pulses are expected to dress the electronic states of a magnetic system and renormalize its exchange energy scales, thus giving rise to a variety of Floquet magnetic phases [see Fig. 4(d)]. The transient exchange interaction can be enhanced or reduced depending on the pump energy, thus altering the finite-momentum spin excitations of the system [224,254–257]. Alternatively, the spin collective modes can also be altered by distorting the lattice and tuning the interspin distance via nonlinear phonon coupling [248-250]. High-resolution trRIXS at selected L edges will map transient changes of the spin fluctuation spectrum in the driven state and experiments targeting a full reconstruction of the transient magnetic Hamiltonian are currently under way. Insulating oxides with high superexchange energies, such as one-dimensional (Sr<sub>2</sub>CuO<sub>3</sub> [29],  $Sr_{14}Cu_{24}O_{41}$  [59]) and two-dimensional cuprates (La<sub>2</sub>CuO<sub>4</sub>) [26]), nickelates ( $La_{2-x}Sr_xNiO_4$  [68], and NdNiO<sub>2</sub> [118]), and iridates ( $Sr_2IrO_4$  [91], and  $Sr_3Ir_2O_7$  [231]) represent a fertile ground for observing these effects. These compounds can be excited within the insulating gap and/or at resonance with high-energy phonons with minimal heating effects, while their magnetic excitation spectrum is accessible with a sub-100-meV energy resolution.

An intriguing target for future trRIXS studies is identifying novel forms of condensation. Photoexcited Mott insulators are predicted to host an entirely new form of condensation, called  $\eta$  pairing [225,226,258–261]. The  $\eta$ -paired phase is a superfluid of doubly occupied electronic states (doublons) with finite momentum  $\mathbf{Q}_n$  and associated with a broken SU(2) symmetry of the Hubbard Hamiltonian. This state is theoretically predicted to emerge in Mott insulators when photodoping holons and doublons by above-gap ultrafast excitation [225,260,261] or by dynamically renormalizing the on-site Coulomb repulsion [262]. The time and energy resolution of the current generation of trRIXS spectrometers is sufficient to test these theories in Cu  $L_3$ - and O K-edges measurements of undoped or doped Mott insulators, such as Sr<sub>2</sub>CuO<sub>3</sub>. If condensation and off-diagonal long-range order are established by the pump to make a state with  $\eta$  pairing, this would induce clearly detectable signatures in the dynamical response. Such signatures include a divergent quasielastic pairing susceptibility at  $Q_n$  [263], three collective modes at energies  $\hbar\omega = 0, \pm (U - 2\mu)$  ( $\mu$  being the chemical potential) [259], and emergent replicas of collective modes in the structure factors [see Fig. 4(f)]. Crucially, this phase is gapless, and therefore challenging to detect with optical spectroscopy. The discovery of  $\eta$  pairing would be groundbreaking as it would unveil a new type of nonequilibrium superconductivity in the absence of a parent equilibrium superconducting phase.

Finally, an important frontier of ultrafast research is represented by the quest to stabilize transient dynamical states and to search for metastable hidden phases [see Fig. 4(e)] that evade thermalization. While several metastable lightdriven phases intimately involve structural rearrangements [209,211,264], purely electronic nonthermal states are rare. These states are called "prethermal" when their decay is prevented by approximate conservation laws or symmetry constraints, "classically metastable" when identifiable with a free-energy minimum of emergent classical variables, or "dissipative steady states" when maintained out of equilibrium by a balance of driving and dissipation [226].

trRIXS could track ultrafast pathways into new hidden or metastable phases and help design these long-lived electronic states. Photoexcited Mott insulators and quantum magnets are promising systems for the observation of prethermal states activated by the selective melting of competing orders or by the realization of metastable electronic distributions. In the first case, trRIXS measurements of spin and dd excitations [265] in photodoped KCuF<sub>3</sub> could directly test the existence of a long-lived hidden state with dominant magnetic orders and flipped orbital pseudospin, which is theoretically predicted to occur in a driven three-quarter filled, two-band  $(d_{y^2-z^2})$  and  $d_{3x^2-r^2}$ ) Hubbard model [266]. Alternatively, tailored pump pulses could transiently break lattice symmetries and activate forbidden hopping pathways between structural units (e.g., a charge reservoir and a metallic plane) that are decoupled at equilibrium. Electrons would form a new quasi-steady-state distribution that is symmetry protected against thermalization after photoexcitation. We anticipate this mechanism to be observable in low-dimensional, undoped and doped Mott insulators (e.g.,  $Sr_{14-x}Ca_xCu_{24}O_{41}$ ), as well as in charge-density wave systems (1T-TaS<sub>2</sub> [267]). trRIXS experiments at the transition metal L edge or at the ligand K edge can directly test for specific transfers of charge into the metastable state and will allow us to accurately determine the nonequilibrium electronic distribution. The longer timescales of these phenomena will allow the use of more aggressive monochromatization schemes which, sustained by the high flux of high-repetition rate XFELs, will enable experiments approaching a 10 meV energy resolution with adequate count rates.

# **VI. FUNCTIONAL MATERIALS**

As described in Sec. I, quantum materials typically involve strong coupling between multiple charge, lattice, orbital, and spin degrees of freedom and high sensitivity to external perturbations such as electric and/or magnetic fields, strain, or photoillumination. These properties offer promising routes toward new functionalities, which may provide the foundation for future technologies [268]. For example, there are extensive efforts to devise methods to switch quantum materials between different electronic phases and/or to use quantum materials to transduce information between electrical, magnetic, or optical forms. Another major theme in this area is to incorporate quantum



FIG. 5. RIXS has unique assets for probing functional materials, such as the structure and propagation of novel magnetic excitons in van der Waals materials. (a) Van der Waals heterostructures interacting with light, electric, and magnetic fields, which offer routes to switching between different electronic phases and the transduction of information. (b) Illustration of Frenkel, Hubbard, and Hund's excitons. (c) RIXS is sensitive to nominally dipole-forbidden excitations and can detect excitons, such as the 1.45 eV feature (circled in white) in the Ni  $L_3$ -edge spectra of NiPS<sub>3</sub> [275].

materials into functional devices. New developments in RIXS setups, coupled with the emergence of new platforms such as magnetic van der Waals materials, suggest that this nascent area may expand significantly in the coming years.

### A. Artificial materials and devices

The ability of RIXS to deliver detailed spectroscopic information from thin layers using a small beam makes it well suited for investigating functional quantum materials in devicelike geometries. During the past decade RIXS has emerged as a powerful probe in these circumstances and has deepened our understanding of how magnetic and electronic phases can be tuned at interfaces in cuprate- [89], iron- [69,90], iridate- [269], nickelate- [270,271], and titanate-based [272] systems.

Magnetic van der Waals materials have recently become a major focus of research [10]. These materials are very diverse and often exhibit strong electron-electron interactions, and they are consequently being intensely studied as a platform for discovering novel correlated states. One particularly important aspect of these materials is their cleavability, which allows for the creation of heterostructures and devicelike configurations. This property enhances our ability to control and investigate nanoscale phases under applied light, electric, and magnetic fields, as shown in Fig. 5(a). Extensive efforts are under way to realize new physics by stacking layers of different van der Waals materials, and leveraging charge transfer, strain, magnetic exchange, and orbital hybridization as additional tuning knobs. However, electronic states at interfaces often change in poorly understood and uncontrolled ways. Thus far, only a handful of materials have exhibited stable excitons and magnetism down to the monolayer limit [273,274]. Understanding how electronic interactions are modified in these heterostructures is thus critical for successfully targeting new phenomena in this space.

Although few-layer-thick exfoliated samples intercept only a small fraction of the x-ray beam, the feasibility of measuring these systems with RIXS has already been demonstrated [90,276]. For example, a recent study has indicated that the charge-transfer energy in NiPS<sub>3</sub> appears to be reduced in few-layer flakes compared to the bulk [276]. Here, the resonant nature of RIXS avoids background signals and allows one to selectively probe different layers within a heterostructure. This area will be bolstered by the delivery of nano-RIXS setups with x-ray beams of order 100 nm [277] and will require parallel technical developments to mitigate new experimental challenges such as beam heating and beam damage. Preliminary work such as mounting samples directly on highly conductive surfaces (e.g., gold) and the use of capping layers made of inert materials has already demonstrated several potential routes to addressing these issues. Studying van der Waals devices in situ under applied electrical, strain, and thermal fields is another frontier challenge. Van der Waals flakes are suitable for preparing pristine, small cross-section samples that can be used to achieve high field density. For the same reasons, these materials are also highly compatible with emerging methods for studying materials under high strain [278].

### **B.** Excitons

Excitons are an emerging topic in RIXS studies of functional materials. These collective excitations play a crucial role in determining the optical properties of quantum materials and have numerous applications in photonics and energy science [279]. Excitons are more likely to form in low-dimensional materials, where the reduced screening of the Coulomb interactions promotes their formation. Hence, van der Waals materials, in either bulk or exfoliated form, are one of the most appropriate platforms for realizing and studying them. One of the most common types of excitons are the Frenkel type, which usually form in weakly correlated semiconductors and are composed of electronhole pairs bound by the screened Coulomb interaction [280,281] [see Fig. 5(b)]. In most cases, Frenkel excitons are well understood using approaches based on the Bethe-Salpeter equation [282]. Excitons in strongly correlated quantum materials are less understood and represent an active research area. In strongly correlated systems, local Coulomb and Hund's exchange interactions play a major role in exciton formation and the exciton can no longer be accurately modeled in terms of the bound states of two interacting single-particle wave functions. As seen in Fig. 5(b), Hubbard and Hund's excitons modify the local spin configuration within the sample, which means that these excitons couple strongly to magnetism. Although manybody excitons exist in several classes of compounds, van der Waals materials have been shown to host unique coupling between excitons and magnetism, including magnetic state-dependent polarized light emission [283,284] and exciton energy [9]. These phenomena offer compelling possibilities for new types of magneto-optical physics and functionalities [285] and so far appear to be unique to van der Waals materials.

A fundamental challenge in studying strongly correlated excitons is that many of them are nominally optical dipole forbidden and the theoretical description of their optical cross section is the subject of active research [282]. Without an established theory for the optical response of correlated excitons, it can be challenging to definitively determine their electronic structure. Since RIXS involves two dipole transitions, excitons can be directly excited in the scattering process [see Fig. 5(c)] as described by the KH equation. Thus, RIXS, combined with theory as described in Sec. VII, provides a powerful means of determining the electronic character of these excitons.

Another special asset of RIXS is its ability to probe the exciton's momentum-dependent dispersion, which will allow high-energy resolution studies to determine how magnetic van der Waals excitons propagate, as has already been done via Ni *L*-edge measurements of NiPS<sub>3</sub>, NiCl<sub>2</sub>, NiBr<sub>2</sub>, and NiI<sub>2</sub> [286,287]. Given that these excitons behave as spin vacancies within the magnetic lattice, it is likely that propagating excitons will be dressed by magnetic excitations. Measurements of exciton dispersion offer the possibility of extracting the detailed exciton-magnon interaction, which underlies modern proposals for exciton-based coherent magnon sensing [9]. A better understanding of magnetic exchange bias in heterostructures that couple exciton-hosting and magnetically ordered layers offers pathways to control the chirality of exciton emission [288,289].

Upon interaction with light, excitonic states hybridize with the electromagnetic field, giving rise to emergent excitonpolariton excitations involving collective oscillations of the polarization in the solid. These states have been recently identified in breakthrough experiments in illuminated magnetic van der Waals materials such as NiPS<sub>3</sub> and CrSBr [284,290,291]. These types of exciton are being heavily investigated in the context of the detection, harvesting,



FIG. 6. Summary of approaches to calculating RIXS: finite cluster, effective cross section, quantum embedding. (a) Finite cluster methods, where the RIXS cross section is evaluated using the exact many-body eigenstates of a small cluster, (b) effective models, where the RIXS cross section is replaced with a suitable approximation, and (c) quantum embedding methods, where the continuum of final-state excitations is captured using a dynamical mean field. Panel (d) shows oxygen *K*-edge RIXS spectra obtained from small cluster models for the cuprate  $La_{1.65}Sr_{0.35}CuO_4$  and low-valence nickelate  $La_4Ni_3O_8$  materials [119]. Panel (e) shows the model phonon excitations  $La_{2-x}Sr_xCuO_4$  compared against the energies of the modes measured at the O *K* edge [146]. Panel (f) shows dynamical mean-field theory (DMFT) predicted Fe *L*-edge spectra for Fe<sub>2</sub>O<sub>3</sub> with a mixture of fluorescencelike and Raman-like excitations [306].

emission, propagation, and modulation of light [292], and coupling RIXS with continuous sample illumination will enable new approaches to studying the spectroscopic properties of exciton polaritons at previously inaccessible momenta.

## **VII. THEORY**

As emphasized in Sec. I C, RIXS spectra are information rich, but some of this information is accessible only through a detailed understanding of the cross section, including the core hole's influence on the system. The field has, consequently, benefited significantly from close collaborations between theory and experiment. In this section, we discuss some of the major theoretical approaches for modeling RIXS experiments, emphasizing what we believe are the remaining challenges and how the field will evolve to tackle them in the coming years.

## A. Finite clusters

Small cluster methods are a direct approach for treating the complexity of the RIXS cross section. This class of methods approximates the infinite system with a finite cluster that is small enough that it can be solved exactly using a suitable nonperturbative numerical method, as illustrated in Fig. 6(a). The KH cross section is then evaluated directly using the exact many-body eigenstates of the cluster while accounting for the system's many-body interactions and the influence of the core hole. Because of the finite extent of the cluster, these methods are appropriate for describing experiments where the length scales of the targeted excitations are comparable to the cluster size. For example, small cluster exact diagonalization (ED) played a pivotal role in understanding local dd [60,293,294] and other orbital or charge excitations [119,293,295–299] in strongly correlated systems [see Fig. 6(d)]. Cluster-based approaches have also been applied to many-body exciton states, as described in Sec. VIB, provided that the spatial extent of the exciton is small enough to be captured within the cluster [275,286,287]. Nowadays, codes for performing these kinds of calculations are quite advanced [300-302] and we envisage that the research landscape will pivot to incorporating these tools into standard experimental workflows, for example, by using machine learning to automatically extract model Hamiltonians from the data and to incorporate real-time theory feedback to optimize data collection during beam times [303–305].

Modeling the excitations of an interacting quantum system requires knowledge of its many-body wave functions. Because the size of the associated Fock space increases factorially in particle number and system size, cluster sizes are generally quite small, resulting in large energy spacing and coarse momentum resolution. For this reason, cluster methods often struggle to capture systems with a continuum of final-state excitations [307] (with the notable exception of quasi-1D systems [156,172,308–310]) or systems with strong electron-phonon coupling [86,87,311,312] (see Sec. VII D). Thus, there is an urgent need to develop more advanced cluster solvers. The future of this topic is likely to focus on co-opting and extending existing advanced many-body numerical methods to calculate RIXS. Recent progress in this direction includes the introduction of density matrix renormalization group (DMRG) based solvers [313,314] for quasi-1D systems. We expect that these approaches will play an increasingly important role in the future, as modern DMRG implementations continue to push into the second dimension [315,316].

Another promising approach to reducing the computational complexity of cluster methods is to explicitly model the photon absorption and emission events in the time domain [314,317–319]. This approach reduces the calculation of the RIXS cross section to time evolving the system during the scattering event, which can be done efficiently using, for example, time-dependent DMRG in the case of a 1D system. Here, the optical absorption process can be treated classically to further reduce the computational complexity of the problem [314]. This approach can also be easily extended to nonequilibrium settings (see Sec. VII E), and will thus play a prominent role in the interpretation of the experiments outlined in Sec. V.

#### **B.** Effective cross sections

While one can express the RIXS intensity as the Fourier transform of a two-particle correlation function, the relevant operators  $O_{\mathbf{q}}(t)$  are complicated and challenging to identify. Their exact form depends on the material being studied, the x-ray edge, and the energy range probed by the photons. Nevertheless, procedures have been developed for systematically formulating effective theories of RIXS in terms of "simpler" correlation functions that can be easier to compute and interpret [see Fig. 6(b)].

An early and highly influential approach along these lines is the ultrashort core-hole lifetime (UCL) expansion. It exploits the fact that the core-hole lifetime is often the shortest timescale in the problem [320], allowing one to expand the KH cross section as a series of multiparticle correlation functions. The lowest-order terms ( $\propto 2/\Gamma$ ) are closely related to the pure charge, spin, and orbital response functions, while higher-order terms involve multiparticle correlation functions describing coupled spin-chargeorbital excitations [197,320,321]. The UCL has been widely applied to various materials but has been most successful in describing low-energy collective excitations in materials whose energy is a small fraction of the inverse core-hole lifetime  $\Gamma/2$ . For example, it played a key role in early studies of magnetic excitations in high- $T_c$  cuprates [26,322].

Effective theories like the UCL provide a direct means to compute RIXS spectra using established many-body methods. For example, while the multiparticle correlation functions may be complex, they can be calculated using methods like quantum Monte Carlo (QMC) or DMRG. So far, studies like these have focused on *L*-edge measurements of systems with nearly complete valence shells (e.g.,  $3d^8$  or  $3d^9$ ). In particular, the low-energy magnetic excitations of the AFM spin-1 chain Y<sub>2</sub>BaNiO<sub>5</sub> probed at the Ni L edge have been described using a combination of dipolar and quadrupolar spin correlation functions computed using DMRG [175], while QMC calculations of the dynamical spin structure factor have been used as proxies for the Cu L-edge spectra of 1D [172,310] and 2D cuprates [323]. However, this approach will face key challenges moving forward. Low-order expansions can be slow to converge to the full KH intensity in a way that depends on the scattering geometry [323]. Therefore, the expansion order will vary for different edges and scattering conditions. A low-order expansion has been derived only for L-edge measurements on  $d^9$  systems [323], and generalizations to new systems or different edges will need to be worked out on a case-by-case basis. In this context, an alternative operator expansion formalism [324], which expands the RIXS operator in terms of local operators acting in the vicinity of the core-hole site, could provide a more direct path forward. This approach directly circumvents the need for a small expansion parameter and dispenses with the requirement of making assumptions about the valence state of the system.

Feynman diagram methods are another emerging approach to formulating effective RIXS theories. Here, the cross section is cast as perturbation series of contributions to the RIXS amplitude, as represented by a series of Feynman diagrams, like the one-phonon diagram shown in Fig. 6(b)[325]. Here, the relevant diagrams are evaluated using suitable propagators. For example, lattice excitations in high- $T_c$  cuprates probed at the Cu L and O K edges [146,325] [see Fig. 6(e)] have been modeled with very high momentum resolution using noninteracting electron and phonon propagators. As with the UCL expansions, this approach can be systematically improved by using dressed electron and phonon propagators or by adding higher-order diagrams. For example, collective spin and charge excitations can be described by replacing the phonon propagator in Fig. 6(b) with the appropriate dynamical susceptibility [326]. Such an approach has already been used to model O K-edge measurements on high- $T_c$  cuprates, where the relevant propagators were derived from cluster perturbation theory [327].

## C. Quantum embedding

While RIXS studies of correlated metals and semimetals are becoming increasing prevalent, many of the finite cluster approaches described above lack the energy and momentum resolution needed to accurately describe their continuum of final states and fluorescence features [270,328]. A current research frontier involves developing quantum embedding methods like dynamical mean-field theory (DMFT) to capture this aspect of the data [see Fig. 6(c)] [270,306,329,330]. Here, the impurity site captures the local correlations and multiplet effects, while the final-state

continuum is encoded in the dynamical mean field. These calculations can thus describe both Raman-like and fluorescencelike excitations in experiments such as that shown in Fig. 6(f). However, a critical and unresolved question relates to how one should compare the predicted spectra to experiments. RIXS is a coherent scattering process, which sums over all possible core-hole sites in the lattice [see Eq. (2)]. Conversely, DMFT methods based on Anderson impurity models only simulate the core hole at the impurity site while leaving the mean-field bath unchanged in the intermediate state. The calculated RIXS response is thus akin to an unmeasurable "local" response, where the core hole is only created at a single site. While one can leverage cluster extensions like dynamical cluster approximation (DCA) to reencode momentum into the problem, wave-function-based cluster solvers are required to directly evaluate Eqs. (1) and (2). ED would limit the cluster sizes and number of bath levels and thus severely limit momentum and energy resolution. Alternatively, spectra could be formulated using the aforementioned operator or UCL expansions, where the corresponding correlation functions would be computed using an embedded cluster framework.

Cumulant expansions of the Green's function also show promise for calculating the RIXS response of correlated metals [331]. This method also offers a key advantage of being easily coupled with *ab initio* methods, allowing one to fully account for the crystal structure's chemical complexity and periodicity. The recent progress in developing accurate correlation-exchange functionals for strongly correlated systems [332,333] also means that these *ab initio*based techniques could be more broadly applied to strongly correlated materials in the future; however, detailed benchmarking against nonperturbative approaches like small cluster ED or DMRG will be necessary.

## **D.** Electron-phonon interactions

Another area where theory has been outpaced by experiments is in understanding how EPC can be probed using RIXS. With improved instrument resolution, experiments now frequently resolve low-energy phonon excitations [32,87,88,139,269,311,334–338]. This development is significant because early theoretical modeling suggests that the **q** dependence of the EPC coupling constant  $q(\mathbf{k}, \mathbf{q})$  can be extracted from the intensities of the phonon excitations [86]. Subsequently, several groups have attempted to map the EPC in materials such as the cuprates [32,83,88,139,337], where the exact role of this interaction remains controversial. Accessing lattice excitations with RIXS also provides new opportunities to examine their interplay with spin, orbital, and charge excitations. Indeed, Franck-Condon-like dressing of local dd excitations [26,334,339,340] and EPC-induced renormalizations of magnetic [87] and electronic [311] energy scales have been inferred from such experiments. These experiments can also provide access to orbital-resolved EPC [87] and couplings far from the Fermi level [338], which are extremely difficult to extract from other experiments.

Modeling lattice excitations in RIXS experiments is extremely challenging and remains a crucial problem for the community. For this reason, most quantitative analysis has focused on Holstein-like interactions and relied on single-site models that neglect electron mobility [86,312]. While these assumptions result in closed-form expressions for the phonon excitations, they drastically oversimplify the problem. More recent studies have found that relaxing these approximations can qualitatively change the results. For example, local coupling to multiple phonon modes [312], changes in local harmonic potentials in the intermediate state [312], reintroducing electron mobility [341], and dispersive phonons [342] all impact the predicted spectra. It is, therefore, highly desirable to develop more robust methods for predicting lattice excitations in RIXS experiments. In this context, the most viable option is likely to be combining diagrammatic approaches [325,326] with many-body methods for efficiently computing electron and phonon Green's functions of EPC Hamiltonians. Modeling of nonlinear EPC and lattice anharmonicity will also be needed to guide future nonequilibirum experiments leveraging nonlinear phononics, where the atoms are significantly displaced from their equilibrium positions.

Finally, some have argued that the lattice responds to the combined charge distribution of the core hole and excited electron in the intermediate state [343,344]. Thus, the EPC probed by RIXS may in reality be an exciton-lattice coupling, with important implications for interpreting and analyzing phonon excitations. One theoretical analysis has shown that the lattice-core-hole coupling can frustrate polaronic effects on the site where the core hole is created. thus enhancing the effects of electron mobility on the spectra [341]. To address this idea, rigorous calculations for the strength of the core-hole-lattice coupling are needed, together with frameworks capable of capturing itinerancy and other effects. In this context, systematic comparisons between theory and experiment on either quasi-1D (e.g., Sr<sub>2</sub>CuO<sub>3</sub> [156] or Li<sub>2</sub>CuO<sub>2</sub> [311]) or quasi-0D materials (e.g.,  $CuB_2O_4$  [339]), where the spectra can be reliably computed with existing DMRG or ED approaches, would be extremely valuable.

## E. Time-resolved RIXS theory

Another emerging and critical area for theory is the development of methods for describing ultrafast pumpprobe RIXS experiments (see also Sec. V) [223,230,345]. Early breakthrough work has been done in deriving a theory of trRIXS using the Keldysh formalism [317]. This approach captures the full dynamics of the problem, including any overlap between the pump and the core-hole excitation, resulting in a four-time correlation function that is numerically expensive to compute. Applications for strongly correlated systems have thus been restricted to small Hubbard clusters and very short timescales [346]. As such, this approach is well suited to processes such as Floquet effects that occur while the laser pulse is interacting with a material [224].

Developing alternative approaches to computing trRIXS that predict the dynamical evolution after pump excitation will be important for future progress. A very promising approach is to explicitly treat the photon absorption and emission events in the scattering process in a time-dependent framework [314,347], which can readily be extended to nonequilibrium RIXS experiments with reduced computational costs. This approach is expected to facilitate simulations that address longer timescales and is already being applied in DMFT-based methodologies [319,348], where they are providing insights into nonequilibirum charge dynamics [319] and lattice excitations [347] in correlated systems. These approaches are particularly useful for using RIXS to detect dynamical changes in the effective Coulomb repulsion after photoexcitation [251,349].

Another important theoretical advance would be a formal systematic derivation of a nonequilibrium version of the UCL or operator expansions for the RIXS cross section. Using such an approach, it may be possible to reduce the four-time correlation function to a series of effective twotime correlation functions that will be easier to compute. While this approach could miss important interference effects when the core-hole lifetime and pump overlap, it may enable calculations to address slow dynamics or situations where energy transfer between different degrees of freedom is crucial.

# VIII. THE FUTURE IS BRIGHT

RIXS has established itself as a powerful probe of quantum materials, and is set to make key contributions to areas ranging from strongly correlated materials (Sec. III), to spin liquids (Sec. IV), to nonequilibrium (Sec. V) and functional (Sec. VI) phases. RIXS is also increasingly influential in other topics in quantum materials research not covered here, including heavy fermions or Kondo lattices, as well as areas such as catalysis and batteries. This diversity is all the more remarkable in light of the relative youth of this technique. For example, the first measurements of magnons date to only about 15 years ago. Now, RIXS is not only measuring magnons, but doing so under extreme conditions of atomically thin layers and ultrafast transient states. The growing maturity of RIXS is also becoming more apparent. For example, many of the complexities of the RIXS cross section are now being exploited to gain new physical insights, rather than being regarded as a challenge for interpreting spectra.

To date, RIXS has proven particularly impactful in studying correlated insulators and their derived phases like strange metals. This is largely because correlated insulators tend to exhibit comparatively simpler, more easily interpretable spectra. One new direction for RIXS research will focus on extending its application to weakly or moderately correlated topological metals. For instance, RIXS could significantly contribute to the emerging field of kagome metals, which exhibit unique flatband, topological, and charge-order behaviors [350]. Notably, some of these materials feature multiple active magnetic ions, which presents opportunities for RIXS to disentangle modes arising from different sublattices. Another promising avenue is the recent identification of a new class of altermagnetic materials, which hold potential for novel applications. Polarization-dependent RIXS might be particularly effective in identifying spin-based magnon band splitting, which is crucial for understanding the novel magnetotransport behaviors in altermagnets.

Looking further into the future, it is clear that major innovation often arises at the interface between different fields. Entanglement plays a pivotal role in quantum materials, making it highly desirable to develop new methods for its manipulation. A frontier challenge lies at the intersection of entanglement metrology and ultrafast techniques: Can we use lasers to transiently enhance quantum entanglement? Tackling the complexities of both fields simultaneously presents a significant challenge, but the potential rewards are compelling. Achieving success in this endeavor could position RIXS as a unique tool to identify novel transient quantum phases. Quantum phase transitions, occurring when a low-temperature ordered phase is suppressed via a nonthermal tuning parameter, have long intrigued the physics community. This fascination stems from the potential for such transitions to create highly entangled phases, potentially leading to new and exotic states. Can similar phenomena can be induced through ultrafast photoexcitation? While photoexcitation can certainly modulate order parameters, a key goal of ultrafast science is to develop optimal control schemes for the on-demand manipulation of light-induced entangled states, such as light-driven superconductors.

The combination of RIXS with applied fields also represents an expanding area within the field of functional materials. A major goal here is to study how materials can be switched to different phases *in situ*. In the future, these *in situ* studies could be expanded to measure the energy gain side of the RIXS spectrum under electric or thermal gradients. This ambitious idea aims to directly identify and study which collective excitations are propagating in response to these field gradients. This type of information would help us understand the processes relevant to spintronics in unprecedented detail.

It should be emphasized that the technology behind RIXS and time-resolved RIXS is still advancing rapidly, pushing the energy resolution toward and below the 10 meV mark, and each improvement in the energy resolution is bound to expand the applicability of RIXS to new materials and to reveal new excitations and physical phenomena. Sub-10-meV has already been demonstrated in a proof-of-principle experiment in the hard x-ray regime, using novel quartz-based flat-crystal optics [351], and spectrometers targeting sub-10-meV in the soft x-ray regime are currently being commissioned [148]. With several orders of magnitude higher photon flux available at new XFEL sources, implementation of such novel spectrometer schemes for quantum materials research is feasible, and we can expect clearer and higher precision access to low-energy quasiparticles and fast transient timescales going forward. The continued refinement of existing technologies combined with the development of ambitious ideas for improved resolution such as XFEL oscillators, critical-angle transmission gratings, and asymmetrically cut Bragg optics [352–354], means that RIXS is set to continue to shine light on many of the most fascinating problems in quantum materials.

# ACKNOWLEDGMENTS

The authors would like to thank all those who have collaborated with us in pushing this research field forward over the years. We acknowledge Jiří Chaloupka, Gábor Halász, Atsushi Hariki, Di-jing Huang, Jungho Kim, Wei-Sheng Lee, Jan Kuneš, Yao Shen, Justina Schlappa, Thorsten Schmitt, Philip Werner, and Yao Wang for their willingness to share data, and we thank Edoardo Baldini, Valentina Bisogni, Tom Devereaux, Joe Dvorak, Martin Eckstein, Paul Evans, Wei-Sheng Lee, Chi-Chang Kao, Daniel Mazzone, and Kejin Zhou for comments on the manuscript. Work by M. M. and M. P. M. D. was supported by the U.S. Department of Energy (DOE). Division of Materials Science, under Contract No. DE-SC0012704. Work by S.J. was supported by the National Science Foundation under Grant No. DMR-1842056. Work by Y.-J. K. was supported by the Natural Science and Engineering Research Council (NSERC) of Canada.

- [1] B. Keimer and J. Moore, *The physics of quantum materials*, Nat. Phys. **13**, 1045 (2017).
- [2] D. Basov, R. Averitt, and D. Hsieh, *Towards properties on demand in quantum materials*, Nat. Mater. 16, 1077 (2017).
- [3] F. Giustino, J. H. Lee, F. Trier, M. Bibes, S. M. Winter, R. Valentí, Y.-W. Son, L. Taillefer, C. Heil, A. I. Figueroa *et al.*, *The 2021 quantum materials roadmap*, J. Phys. Mater. **3**, 042006 (2021).
- [4] B. Keimer, S. A. Kivelson, M. R. Norman, S. Uchida, and J. Zaanen, From quantum matter to high-temperature superconductivity in copper oxides, Nature (London) 518, 179 (2015).
- [5] P. W. Phillips, N. E. Hussey, and P. Abbamonte, *Stranger than metals*, Science 377, eabh4273 (2022).
- [6] C. Broholm, R. Cava, S. Kivelson, D. Nocera, M. Norman, and T. Senthil, *Quantum spin liquids*, Science 367, 263 (2020).

- [7] A. de la Torre, D. M. Kennes, M. Claassen, S. Gerber, J. W. McIver, and M. A. Sentef, *Colloquium: Nonthermal pathways to ultrafast control in quantum materials*, Rev. Mod. Phys. **93**, 041002 (2021).
- [8] J. Bloch, A. Cavalleri, V. Galitski, M. Hafezi, and A. Rubio, *Strongly correlated electron-photon systems*, Nature (London) **606**, 41 (2022).
- [9] Y. J. Bae, J. Wang, A. Scheie, J. Xu, D. G. Chica, G. M. Diederich, J. Cenker, M. E. Ziebel, Y. Bai, H. Ren, C. R. Dean, M. Delor, X. Xu, X. Roy, A. D. Kent, and X. Zhu, *Exciton-coupled coherent magnons in a 2D semiconductor*, Nature (London) **609**, 282 (2022).
- [10] K. S. Burch, D. Mandrus, and J.-G. Park, *Magnetism in two-dimensional van der Waals materials*, Nature (London) 563, 47 (2018).
- [11] D. Pines, *Elementary Excitations in Solids* (CRC Press, Boca Raton, FL, 1999).
- [12] M. L. Cohen, Looking back and ahead at condensed matter physics, Phys. Today 59, No. 6, 48 (2006).
- [13] D. N. Basov, R. D. Averitt, D. van der Marel, M. Dressel, and K. Haule, *Electrodynamics of correlated electron materials*, Rev. Mod. Phys. 83, 471 (2011).
- [14] T. P. Devereaux and R. Hackl, *Inelastic light scattering from correlated electrons*, Rev. Mod. Phys. **79**, 175 (2007).
- [15] For a detailed derivation of this expression from the lightmatter interaction, see Ament *et al.* [16].
- [16] L. J. P. Ament, M. van Veenendaal, T. P. Devereaux, J. P. Hill, and J. van den Brink, *Resonant inelastic x-ray* scattering studies of elementary excitations, Rev. Mod. Phys. 83, 705 (2011).
- [17] G. L. Squires, Introduction to the Theory of Thermal Neutron Scattering, 3rd ed. (Cambridge University Press, Cambridge, England, 2012).
- [18] A. Q. R. Baron, *Introduction to high-resolution inelastic x-ray scattering*, arXiv:1504.01098.
- [19] W. Schuelke, *Electron Dynamics by Inelastic X-Ray Scattering* (Oxford University Press, New York, 2007).
- [20] H. Ibach and D. L. Mills, *Electron Energy Loss Spectros-copy and Surface Vibrations* (Academic Press, New York, 2013).
- [21] P. Abbamonte and J. Fink, *Collective charge excitations* studied by electron energy-loss spectroscopy, arXiv:2404 .04670.
- [22] D. Pines, *Theory Of Quantum Liquids* (CRC Press, Boca Raton, FL, 1989).
- [23] S. Vig, A. Kogar, M. Mitrano, A. A. Husain, V. Mishra, M. S. Rak, L. Venema, P. D. Johnson, G. D. Gu, E. Fradkin, M. R. Norman, and P. Abbamonte, *Measurement* of the dynamic charge response of materials using low-energy, momentum-resolved electron energy-loss spectroscopy (M-EELS), SciPost Phys. 3, 026 (2017).
- [24] A. Damascelli, Z. Hussain, and Z.-X. Shen, Angle-resolved photoemission studies of the cuprate superconductors, Rev. Mod. Phys. 75, 473 (2003).
- [25] J. A. Sobota, Y. He, and Z.-X. Shen, *Angle-resolved photoemission studies of quantum materials*, Rev. Mod. Phys. **93**, 025006 (2021).
- [26] L. Braicovich, J. van den Brink, V. Bisogni, M. M. Sala, L. J. P. Ament, N. B. Brookes, G. M. De Luca, M. Salluzzo, T. Schmitt, V. N. Strocov, and G. Ghiringhelli,

Magnetic excitations and phase separation in the underdoped  $La_{2-x}Sr_xCuO_4$  superconductor measured by resonant inelastic x-ray scattering, Phys. Rev. Lett. **104**, 077002 (2010).

- [27] J. P. Hill, G. Blumberg, Y.-J. Kim, D. S. Ellis, S. Wakimoto, R. J. Birgeneau, S. Komiya, Y. Ando, B. Liang, R. L. Greene, D. Casa, and T. Gog, *Observation of a 500 meV collective mode in* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub> *and* Nd<sub>2</sub>CuO<sub>4</sub> *using resonant inelastic x-ray scattering*, Phys. Rev. Lett. **100**, 097001 (2008).
- [28] L. J. P. Ament, G. Ghiringhelli, M. M. Sala, L. Braicovich, and J. van den Brink, *Theoretical demonstration of how the dispersion of magnetic excitations in cuprate compounds can be determined using resonant inelastic x-ray scattering*, Phys. Rev. Lett. **103**, 117003 (2009).
- [29] J. Schlappa, K. Wohlfeld, K. J. Zhou, M. Mourigal, M. W. Haverkort, V. N. Strocov, L. Hozoi, C. Monney, S. Nishimoto, S. Singh, A. Revcolevschi, J. S. Caux, L. Patthey, H. M. Rønnow, J. van den Brink, and T. Schmitt, *Spin-orbital separation in the quasi-one-dimensional Mott insulator* Sr<sub>2</sub>CuO<sub>3</sub>, Nature (London) **485**, 82 (2012).
- [30] J. Li, Y. Gu, Y. Takahashi, K. Higashi, T. Kim, Y. Cheng, F. Yang, J. Kuneš, J. Pelliciari, A. Hariki, and V. Bisogni, *Single- and multimagnon dynamics in antiferromagnetic* α-Fe<sub>2</sub>O<sub>3</sub> *thin films*, Phys. Rev. X 13, 011012 (2023).
- [31] H. Elnaggar, A. Nag, M. W. Haverkort, M. Garcia-Fernandez, A. Walters, R.-P. Wang, K.-J. Zhou, and F. de Groot, *Magnetic excitations beyond the singleand double-magnons*, Nat. Commun. 14, 2749 (2023).
- [32] L. Chaix, G. Ghiringhelli, Y. Y. Peng, M. Hashimoto, B. Moritz, K. Kummer, N. B. Brookes, Y. He, S. Chen, S. Ishida, Y. Yoshida, H. Eisaki, M. Salluzzo, L. Braicovich, Z.-X. Shen, T. P. Devereaux, and W.-S. Lee, *Dispersive charge density wave excitations in*  $Bi_2Sr_2CaCu_2O_{8+\delta}$ , Nat. Phys. **13**, 952 (2017).
- [33] G. B. Halász, N. B. Perkins, and J. van den Brink, Resonant inelastic x-ray scattering response of the Kitaev honeycomb model, Phys. Rev. Lett. 117, 127203 (2016).
- [34] M. Lundberg and P. Wernet, Resonant inelastic x-ray scattering (RIXS) studies in chemistry: Present and future, in Synchrotron Light Sources and Free-Electron Lasers: Accelerator Physics, Instrumentation and Science Applications, edited by E. Jaeschke, S. Khan, J. R. Schneider, and J. B. Hastings (Springer International Publishing, 2019), pp. 1–52.
- [35] F. Gel'mukhanov, M. Odelius, S. P. Polyutov, A. Föhlisch, and V. Kimberg, *Dynamics of resonant x-ray and Auger* scattering, Rev. Mod. Phys. **93**, 035001 (2021).
- [36] J. Dvorak, I. Jarrige, V. Bisogni, S. Coburn, and W. Leonhardt, *Towards 10 meV resolution: The design of an ultrahigh resolution soft x-ray RIXS spectrometer*, Rev. Sci. Instrum. 87, 115109 (2016).
- [37] N. Brookes, F. Yakhou-Harris, K. Kummer, A. Fondacaro, J. Cezar, D. Betto, E. Velez-Fort, A. Amorese, G. Ghiringhelli, L. Braicovich, R. Barrett, G. Berruyer, F. Cianciosi, L. Eybert, P. Marion, P. van der Linden, and L. Zhang, *The beamline ID32 at the ESRF for soft x-ray high energy resolution resonant inelastic x-ray scattering and polarisation dependent x-ray absorption spectroscopy*, Nucl. Instrum. Methods Phys. Res., Sect. A **903**, 175 (2018).

- [38] K.-J. Zhou, A. Walters, M. Garcia-Fernandez, T. Rice, M. Hand, A. Nag, J. Li, S. Agrestini, P. Garland, H. Wang, S. Alcock, I. Nistea, B. Nutter, N. Rubies, G. Knap, M. Gaughran, F. Yuan, P. Chang, J. Emmins, and G. Howell, *121: An advanced high-resolution resonant inelastic x-ray scattering beamline at Diamond Light Source*, J. Synchrotron Radiat. 29, 563 (2022).
- [39] A. Singh, H. Y. Huang, Y. Y. Chu, C. Y. Hua, S. W. Lin, H. S. Fung, H. W. Shiu, J. Chang, J. H. Li, J. Okamoto et al., Development of the soft x-ray AGM-AGS RIXS beamline at the Taiwan Photon Source, J. Synchrotron Radiat. 28, 977 (2021).
- [40] H. Gretarsson, D. Ketenoglu, M. Harder, S. Mayer, F.-U. Dill, M. Spiwek, H. Schulte-Schrepping, M. Tischer, H.-C. Wille, B. Keimer, and H. Yavaş, *IRIXS:* A resonant inelastic x-ray scattering instrument dedicated to x-rays in the intermediate energy range, J. Synchrotron Radiat. 27, 538 (2020).
- [41] J. Bertinshaw, S. Mayer, F.-U. Dill, H. Suzuki, O. Leupold, A. Jafari, I. Sergueev, M. Spiwek, A. Said, E. Kasman, X. Huang, B. Keimer, and H. Gretarsson, *IRIXS spectrograph: An ultra high-resolution spectrometer for tender RIXS*, J. Synchrotron Radiat. 28, 1184 (2021).
- [42] H. Suzuki, L. Wang, J. Bertinshaw, H. U. R. Strand, S. Käser, M. Krautloher, Z. Yang, N. Wentzell, O. Parcollet, F. Jerzembeck, N. Kikugawa, A. P. Mackenzie, A. Georges, P. Hansmann, H. Gretarsson, and B. Keimer, *Distinct spin and orbital dynamics in* Sr<sub>2</sub>RuO<sub>4</sub>, Nat. Commun. 14, 7042 (2023).
- [43] P. Eisenberger, P. M. Platzman, and H. Winick, X-ray resonant Raman scattering: Observation of characteristic radiation narrower than the lifetime width, Phys. Rev. Lett. 36, 623 (1976).
- [44] P. Eisenberger, P. M. Platzman, and H. Winick, *Resonant x-ray Raman scattering studies using synchrotron radia-tion*, Phys. Rev. B 13, 2377 (1976).
- [45] J.-E. Rubensson, D. Mueller, R. Shuker, D. L. Ederer, C. H. Zhang, J. Jia, and T. A. Callcott, *Excitation-energy dependence in the* L<sub>2,3</sub> *fluorescence spectrum of* Si, Phys. Rev. Lett. **64**, 1047 (1990).
- [46] Y. Ma, N. Wassdahl, P. Skytt, J. Guo, J. Nordgren, P. D. Johnson, J.-E. Rubensson, T. Boske, W. Eberhardt, and S. D. Kevan, *Soft-x-ray resonant inelastic scattering at the C K edge of diamond*, Phys. Rev. Lett. **69**, 2598 (1992).
- [47] A. Kotani and S. Shin, *Resonant inelastic x-ray scattering spectra for electrons in solids*, Rev. Mod. Phys. **73**, 203 (2001).
- [48] C.-C. Kao, W. A. L. Caliebe, J. B. Hastings, and J.-M. Gillet, X-ray resonant Raman scattering in NiO: Resonant enhancement of the charge-transfer excitations, Phys. Rev. B 54, 16361 (1996).
- [49] L. Braicovich, C. Dallera, G. Ghiringhelli, N. B. Brookes, J. B. Goedkoop, and M. A. van Veenendaal, X-ray L<sub>2,3</sub> resonant Raman scattering from NiO: Spin flip and intermediate-state relaxation, Phys. Rev. B 55, R15989 (1997).
- [50] F. M. F. de Groot, P. Kuiper, and G. A. Sawatzky, *Local spin-flip spectral distribution obtained by resonant x-ray Raman scattering*, Phys. Rev. B 57, 14584 (1998).
- [51] S. M. Butorin, D. C. Mancini, J.-H. Guo, N. Wassdahl, J. Nordgren, M. Nakazawa, S. Tanaka, T. Uozumi, A. Kotani,

Y. Ma, K. E. Myano, B. A. Karlin, and D. K. Shuh, *Resonant x-ray fluorescence spectroscopy of correlated systems: A probe of charge-transfer excitations*, Phys. Rev. Lett. **77**, 574 (1996).

- [52] M. P. M. Dean, Insights into the high temperature superconducting cuprates from resonant inelastic x-ray scattering, J. Magn. Magn. Mater. 376, 3 (2015).
- [53] J. P. Hill, C.-C. Kao, W. A. L. Caliebe, M. Matsubara, A. Kotani, J. L. Peng, and R. L. Greene, *Resonant inelastic x-ray scattering in* Nd<sub>2</sub>CuO<sub>4</sub>, Phys. Rev. Lett. **80**, 4967 (1998).
- [54] P. Abbamonte, C. A. Burns, E. D. Isaacs, P. M. Platzman, L. L. Miller, S. W. Cheong, and M. V. Klein, *Resonant inelastic x-ray scattering from valence excitations in insulating copper oxides*, Phys. Rev. Lett. 83, 860 (1999).
- [55] M.Z. Hasan, E.D. Isaacs, Z.-X. Shen, L.L. Miller, K. Tsutsui, T. Tohyama, and S. Maekawa, *Electronic structure of Mott insulators studied by inelastic x-ray scattering*, Science 288, 1811 (2000).
- [56] Y. J. Kim, J. P. Hill, C. A. Burns, S. Wakimoto, R. J. Birgeneau, D. Casa, T. Gog, and C. T. Venkataraman, *Resonant inelastic x-ray scattering study of charge excitations in* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub>, Phys. Rev. Lett. **89**, 177003 (2002).
- [57] G. Ghiringhelli, N. B. Brookes, E. Annese, H. Berger, C. Dallera, M. Grioni, L. Perfetti, A. Tagliaferri, and L. Braicovich, Low energy electronic excitations in the layered cuprates studied by copper L<sub>3</sub> resonant inelastic x-ray scattering, Phys. Rev. Lett. **92**, 117406 (2004).
- [58] Y. Shvyd'ko, J. Hill, C. Burns, D. Coburn, B. Brajuskovic, D. Casa, K. Goetze, T. Gog, R. Khachatryan, J.-H. Kim, C. Kodituwakku, M. Ramanathan, T. Roberts, A. Said, H. Sinn, D. Shu, S. Stoupin, M. Upton, M. Wieczorek, and H. Yavas, *MERIX—Next generation medium energy resolution inelastic x-ray scattering instrument at the APS*, J. Electron Spectrosc. Relat. Phenom. **188**, 140 (2013).
- [59] J. Schlappa, T. Schmitt, F. Vernay, V. N. Strocov, V. Ilakovac, B. Thielemann, H. M. Rønnow, S. Vanishri, A. Piazzalunga, X. Wang, L. Braicovich, G. Ghiringhelli, C. Marin, J. Mesot, B. Delley, and L. Patthey, *Collective* magnetic excitations in the spin ladder Sr<sub>14</sub>Cu<sub>24</sub>O<sub>41</sub> measured using high-resolution resonant inelastic x-ray scattering, Phys. Rev. Lett. **103**, 047401 (2009).
- [60] G. Ghiringhelli, A. Piazzalunga, C. Dallera, T. Schmitt, V. N. Strocov, J. Schlappa, L. Patthey, X. Wang, H. Berger, and M. Grioni, *Observation of two nondispersive magnetic excitations in NiO by resonant inelastic soft-x-ray scattering*, Phys. Rev. Lett. **102**, 027401 (2009).
- [61] M. Guarise, B. Dalla Piazza, M. Moretti Sala, G. Ghiringhelli, L. Braicovich, H. Berger, J. N. Hancock, D. van der Marel, T. Schmitt, V. N. Strocov, L. J. P. Ament, J. van den Brink, P.-H. Lin, P. Xu, H. M. Rønnow, and M. Grioni, *Measurement of magnetic excitations in the two-dimensional antiferromagnetic* Sr<sub>2</sub>CuO<sub>2</sub>Cl<sub>2</sub> *insulator using resonant x-ray scattering: Evidence for extended interactions*, Phys. Rev. Lett. **105**, 157006 (2010).
- [62] M. Le Tacon, G. Ghiringhelli, J. Chaloupka, M. M. Sala, V. Hinkov, M. W. Haverkort, M. Minola, M. Bakr, K. J. Zhou, S. Blanco-Canosa, C. Monney, Y. T. Song, G. L. Sun, C. T. Lin, G. M. De Luca, M. Salluzzo, G. Khaliullin,

T. Schmitt, L. Braicovich, and B. Keimer, *Intense para-magnon excitations in a large family of high-temperature superconductors*, Nat. Phys. **7**, 725 (2011).

- [63] M. P. M. Dean, A. J. A. James, R. S. Springell, X. Liu, C. Monney, K. J. Zhou, R. M. Konik, J. S. Wen, Z. J. Xu, G. D. Gu, V. N. Strocov, T. Schmitt, and J. P. Hill, *Highenergy magnetic excitations in the cuprate superconductor*  $Bi_2Sr_2CaCu_2O_{8+\delta}$ : *Towards a unified description of its electronic and magnetic degrees of freedom*, Phys. Rev. Lett. **110**, 147001 (2013).
- [64] M. P. M. Dean, G. Dellea, R. S. Springell, F. Yakhou-Harris, K. Kummer, N. B. Brookes, X. Liu, Y.-J. Sun, J. Strle, T. Schmitt, L. Braicovich, G. Ghiringhelli, I. Božović, and J. P. Hill, *Persistence of magnetic excitations in* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub> *from the undoped insulator to the heavily overdoped non-superconducting metal*, Nat. Mater. **12**, 1019 (2013).
- [65] W. S. Lee, J. J. Lee, E. A. Nowadnick, S. Gerber, W. Tabis, S. W. Huang, V. N. Strocov, E. M. Motoyama, G. Yu, B. Moritz, H. Y. Huang, R. P. Wang, Y. B. Huang, W. B. Wu, C. T. Chen, D. J. Huang, M. Greven, T. Schmitt, Z. X. Shen, and T. P. Devereaux, Asymmetry of collective excitations in electron- and hole-doped cuprate superconductors, Nat. Phys. 10, 883 (2014).
- [66] K. Ishii, M. Fujita, T. Sasaki, M. Minola, G. Dellea, C. Mazzoli, K. Kummer, G. Ghiringhelli, L. Braicovich, T. Tohyama, K. Tsutsumi, K. Sato, R. Kajimoto, K. Ikeuchi, K. Yamada, M. Yoshida, M. Kurooka, and J. Mizuki, *Highenergy spin and charge excitations in electron-doped copper oxide superconductors*, Nat. Commun. 5, 3714 (2014).
- [67] K.-J. Zhou, Y.-B. Huang, C. Monney, X. Dai, V.N. Strocov, N.-L. Wang, Z.-G. Chen, C. Zhang, P. Dai, L. Patthey, J. van den Brink, H. Ding, and T. Schmitt, *Persistent high-energy spin excitations in iron-pnictide superconductors*, Nat. Commun. 4, 1 (2013).
- [68] G. Fabbris, D. Meyers, L. Xu, V. M. Katukuri, L. Hozoi, X. Liu, Z.-Y. Chen, J. Okamoto, T. Schmitt, A. Uldry, B. Delley, G. D. Gu, D. Prabhakaran, A. T. Boothroyd, J. van den Brink, D. J. Huang, and M. P. M. Dean, *Doping dependence of collective spin and orbital excitations in the spin-1 quantum antiferromagnet* La<sub>2-x</sub>Sr<sub>x</sub>NiO<sub>4</sub> *observed by x rays*, Phys. Rev. Lett. **118**, 156402 (2017).
- [69] J. Pelliciari, S. Lee, K. Gilmore, J. Li, Y. Gu, A. Barbour, I. Jarrige, C. H. Ahn, F. J. Walker, and V. Bisogni, *Tuning spin excitations in magnetic films by confinement*, Nat. Mater. 20, 188 (2021).
- [70] J. Kim, D. Casa, M. H. Upton, T. Gog, Y.-J. Kim, J. F. Mitchell, M. van Veenendaal, M. Daghofer, J. van den Brink, G. Khaliullin, and B. J. Kim, *Magnetic excitation spectra of* Sr<sub>2</sub>IrO<sub>4</sub> *probed by resonant inelastic x-ray scattering: Establishing links to cuprate superconductors*, Phys. Rev. Lett. **108**, 177003 (2012).
- [71] J. Kim, A. H. Said, D. Casa, M. H. Upton, T. Gog, M. Daghofer, G. Jackeli, J. van den Brink, G. Khaliullin, and B. J. Kim, Large spin-wave energy gap in the bilayer iridate Sr<sub>3</sub>Ir<sub>2</sub>O<sub>7</sub>: Evidence for enhanced dipolar interactions near the Mott metal-insulator transition, Phys. Rev. Lett. **109**, 157402 (2012).

- [72] W.-G. Yin, X. Liu, A. M. Tsvelik, M. P. M. Dean, M. H. Upton, J. Kim, D. Casa, A. Said, T. Gog, T. F. Qi, G. Cao, and J. P. Hill, *Ferromagnetic exchange anisotropy from antiferromagnetic superexchange in the mixed 3d 5d transition-metal compound* Sr<sub>3</sub>CuIrO<sub>6</sub>, Phys. Rev. Lett. **111**, 057202 (2013).
- [73] H. Gretarsson, J. P. Clancy, X. Liu, J. P. Hill, E. Bozin, Y. Singh, S. Manni, P. Gegenwart, J. Kim, A. H. Said, D. Casa, T. Gog, M. H. Upton, H.-S. Kim, J. Yu, V. M. Katukuri, L. Hozoi, J. van den Brink, and Y.-J. Kim, *Crystal-field splitting and correlation effect on the electronic structure of A*<sub>2</sub>IrO<sub>3</sub>, Phys. Rev. Lett. **110**, 076402 (2013).
- [74] J. Kim, M. Daghofer, A. Said, T. Gog, J. Van den Brink, G. Khaliullin, and B. Kim, *Excitonic quasiparticles in a spinorbit Mott insulator*, Nat. Commun. 5, 1 (2014).
- [75] S. Hwan Chun, J.-W. Kim, J. Kim, H. Zheng, C. C. Stoumpos, C. D. Malliakas, J. F. Mitchell, K. Mehlawat, Y. Singh, Y. Choi, T. Gog, A. Al-Zein, M. M. Sala, M. Krisch, J. Chaloupka, G. Jackeli, G. Khaliullin, and B. J. Kim, *Direct evidence for dominant bond-directional interactions in a honeycomb lattice iridate* Na<sub>2</sub>IrO<sub>3</sub>, Nat. Phys. 11, 462 (2015).
- [76] B. J. Kim, H. Jin, S. J. Moon, J.-Y. Kim, B.-G. Park, C. S. Leem, J. Yu, T. W. Noh, C. Kim, S.-J. Oh, J.-H. Park, V. Durairaj, G. Cao, and E. Rotenberg, *Novel J<sub>eff</sub> = 1/2 Mott state induced by relativistic spin-orbit coupling in* Sr<sub>2</sub>IrO<sub>4</sub>, Phys. Rev. Lett. **101**, 076402 (2008).
- [77] G. Jackeli and G. Khaliullin, Mott insulators in the strong spin-orbit coupling limit: From Heisenberg to a quantum compass and Kitaev models, Phys. Rev. Lett. 102, 017205 (2009).
- [78] G. Ghiringhelli, M. L. Tacon, M. Minola, S. Blanco-Canosa, C. Mazzoli, N. B. Brookes, G. M. D. Luca, A. Frano, D. G. Hawthorn, F. He, T. Loew, M. M. Sala, D. C. Peets, M. Salluzzo, E. Schierle, R. Sutarto, G. A. Sawatzky, E. Weschke, B. Keimer, and L. Braicovich, *Long-range incommensurate charge fluctuations in* (*Y*,*Nd*)Ba<sub>2</sub>Cu<sub>3</sub>O<sub>6+x</sub>, Science **337**, 821 (2012).
- [79] W. Tabis, Y. Li, M. L. Tacon, L. Braicovich, A. Kreyssig, M. Minola, G. Dellea, E. Weschke, M. J. Veit, M. Ramazanoglu, A. I. Goldman, T. Schmitt, G. Ghiringhelli, N. Barišić, M. K. Chan, C. J. Dorow, G. Yu, X. Zhao, B. Keimer, and M. Greven, *Charge order and its connection with Fermi-liquid charge transport in a pristine high-T<sub>c</sub> cuprate*, Nat. Commun. **5**, 5875 (2014).
- [80] H. Miao, J. Lorenzana, G. Seibold, Y.Y. Peng, A. Amorese, F. Yakhou-Harris, K. Kummer, N. B. Brookes, R. M. Konik, V. Thampy, G. D. Gu, G. Ghiringhelli, L. Braicovich, and M. P. M. Dean, *High-temperature charge density wave correlations in* La<sub>1.875</sub>Ba<sub>0.125</sub>CuO<sub>4</sub> *without spin-charge locking*, Proc. Natl. Acad. Sci. U.S.A. **114**, 12430 (2017).
- [81] Y. Y. Peng, R. Fumagalli, Y. Ding, M. Minola, S. Caprara, D. Betto, M. Bluschke, G. M. De Luca, K. Kummer, E. Lefrançois, M. Salluzzo, H. Suzuki, M. Le Tacon, X. J. Zhou, N. B. Brookes, B. Keimer, L. Braicovich, M. Grilli, and G. Ghiringhelli, *Re-entrant charge order in overdoped* (Bi, Pb)<sub>2.12</sub>Sr<sub>1.88</sub>CuO<sub>6+ $\delta$ </sub> outside the pseudogap regime, Nat. Mater. **17**, 697 (2018).

- [82] R. Arpaia, S. Caprara, R. Fumagalli, G. D. Vecchi, Y. Y. Peng, E. Andersson, D. Betto, G. M. D. Luca, N. B. Brookes, F. Lombardi, M. Salluzzo, L. Braicovich, C. D. Castro, M. Grilli, and G. Ghiringhelli, *Dynamical charge density fluctuations pervading the phase diagram of a Cu-based high-T<sub>c</sub> superconductor*, Science **365**, 906 (2019).
- [83] J. Q. Lin, H. Miao, D. G. Mazzone, G. D. Gu, A. Nag, A. C. Walters, M. García-Fernández, A. Barbour, J. Pelliciari, I. Jarrige, M. Oda, K. Kurosawa, N. Momono, K.-J. Zhou, V. Bisogni, X. Liu, and M. P. M. Dean, *Strongly correlated charge density wave in* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub> *evidenced by doping-dependent phonon anomaly*, Phys. Rev. Lett. **124**, 207005 (2020).
- [84] B. Yu, W. Tabis, I. Bialo, F. Yakhou, N. B. Brookes, Z. Anderson, Y. Tang, G. Yu, and M. Greven, *Unusual dynamic charge correlations in simple-tetragonal* HgBa<sub>2</sub>CuO<sub>4+ $\delta$ </sub>, Phys. Rev. X **10**, 021059 (2020).
- [85] C. Tam, M. Zhu, J. Ayres, K. Kummer, F. Yakhou-Harris, J. Cooper, A. Carrington, and S. Hayden, *Charge density* waves and Fermi surface reconstruction in the clean overdoped cuprate superconductor  $Tl_2Ba_2CuO_{6+\delta}$ , Nat. Commun. **13**, 1 (2022).
- [86] L. J. P. Ament, M. van Veenendaal, and J. van den Brink, Determining the electron-phonon coupling strength from resonant inelastic x-ray scattering at transition metal L-edges, Europhys. Lett. 95, 27008 (2011).
- [87] W. S. Lee, S. Johnston, B. Moritz, J. Lee, M. Yi, K. J. Zhou, T. Schmitt, L. Patthey, V. Strocov, K. Kudo, Y. Koike, J. van den Brink, T. P. Devereaux, and Z. X. Shen, *Role of lattice coupling in establishing electronic and magnetic properties in quasi-one-dimensional cuprates*, Phys. Rev. Lett. **110**, 265502 (2013).
- [88] M. Rossi, R. Arpaia, R. Fumagalli, M. Moretti Sala, D. Betto, K. Kummer, G. M. De Luca, J. van den Brink, M. Salluzzo, N. B. Brookes, L. Braicovich, and G. Ghiringhelli, *Experimental determination of momentumresolved electron-phonon coupling*, Phys. Rev. Lett. **123**, 027001 (2019).
- [89] M. P. M. Dean, R. S. Springell, C. Monney, K. J. Zhou, J. Pereiro, I. Božović, B. Dalla Piazza, H. M. Rønnow, E. Morenzoni, J. van den Brink, T. Schmitt, and J. P. Hill, *Spin excitations in a single* La<sub>2</sub>CuO<sub>4</sub> *layer*, Nat. Mater. 11, 850 (2012).
- [90] J. Pelliciari, S. Karakuzu, Q. Song, R. Arpaia, A. Nag, M. Rossi, J. Li, T. Yu, X. Chen, R. Peng, M. García-Fernández, A. C. Walters, Q. Wang, J. Zhao, G. Ghiringhelli, D. Feng, T. A. Maier, K.-J. Zhou, S. Johnston, and R. Comin, *Evolution of spin excitations from bulk to monolayer* FeSe, Nat. Commun. **12**, 3122 (2021).
- [91] M. P. M. Dean, Y. Cao, X. Liu, S. Wall, D. Zhu, R. Mankowsky, V. Thampy, X. M. Chen, J. G. Vale, D. Casa et al., Ultrafast energy- and momentum-resolved dynamics of magnetic correlations in the photo-doped Mott insulator Sr<sub>2</sub>IrO<sub>4</sub>, Nat. Mater. 15, 601 (2016).
- [92] M. Mitrano, S. Lee, A. A. Husain, L. Delacretaz, M. Zhu, G. de la Peña Munoz, S. X.-L. Sun, Y. I. Joe, A. H. Reid, S. F. Wandel, G. Coslovich, W. Schlotter, T. van Driel,

PHYS. REV. X 14, 040501 (2024)

J. Schneeloch, G. D. Gu, S. Hartnoll, N. Goldenfeld, and P. Abbamonte, *Ultrafast time-resolved x-ray scattering reveals diffusive charge order dynamics in*  $La_{2-x}Ba_xCuO_4$ , Sci. Adv. **5**, eaax3346 (2019).

- [93] S. Paschen and Q. Si, *Quantum phases driven by strong correlations*, Nat. Rev. Phys. **3**, 9 (2021).
- [94] H. Takagi, B. Batlogg, H. L. Kao, J. Kwo, R. J. Cava, J. J. Krajewski, and W. F. Peck, *Systematic evolution of temperature-dependent resistivity in* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub>, Phys. Rev. Lett. **69**, 2975 (1992).
- [95] K. T. N. E. Hussey and H. Takagi, Universality of the Mott-Ioffe-Regel limit in metals, Philos. Mag. 84, 2847 (2004).
- [96] N. E. Hussey, A. P. Mackenzie, J. R. Cooper, Y. Maeno, S. Nishizaki, and T. Fujita, *Normal-state magnetoresistance of* Sr<sub>2</sub>RuO<sub>4</sub>, Phys. Rev. B 57, 5505 (1998).
- [97] J. G. Analytis, H. Kuo, R. D. McDonald, M. Wartenbe, P. Rourke, N. Hussey, and I. Fisher, *Transport near a quantum critical point in*  $BaFe_2(As_{1-x}P_x)_2$ , Nat. Phys. **10**, 194 (2014).
- [98] Y. Cao, D. Chowdhury, D. Rodan-Legrain, O. Rubies-Bigorda, K. Watanabe, T. Taniguchi, T. Senthil, and P. Jarillo-Herrero, *Strange metal in magic-angle graphene* with near Planckian dissipation, Phys. Rev. Lett. **124**, 076801 (2020).
- [99] M. Hepting, L. Chaix, E. W. Huang, R. Fumagalli, Y. Y. Peng, B. Moritz, K. Kummer, N. B. Brookes, W. C. Lee, M. Hashimoto, T. Sarkar, J.-F. He, C. R. Rotundu, Y. S. Lee, R. L. Greene, L. Braicovich, G. Ghiringhelli, Z. X. Shen, T. P. Devereaux, and W. S. Lee, *Three-dimensional collective charge excitations in electron-doped copper oxide superconductors*, Nature (London) 563, 374 (2018).
- [100] A. A. Patel, H. Guo, I. Esterlis, and S. Sachdev, Universal theory of strange metals from spatially random interactions, Science 381, 790 (2023).
- [101] C. M. Varma, Colloquium: Linear in temperature resistivity and associated mysteries including high temperature superconductivity, Rev. Mod. Phys. 92, 031001 (2020).
- [102] I. Bozovic, Plasmons in cuprate superconductors, Phys. Rev. B 42, 1969 (1990).
- [103] J. Levallois, M.K. Tran, D. Pouliot, C.N. Presura, L. H. Greene, J. N. Eckstein, J. Uccelli, E. Giannini, G. D. Gu, A. J. Leggett, and D. van der Marel, *Temperature-dependent ellipsometry measurements of partial Coulomb energy in superconducting cuprates*, Phys. Rev. X 6, 031027 (2016).
- [104] J. Lin, J. Yuan, K. Jin, Z. Yin, G. Li, K.-J. Zhou, X. Lu, M. Dantz, T. Schmitt, H. Ding, H. Guo, M. P. M. Dean, and X. Liu, *Doping evolution of the charge excitations and electron correlations in electron-doped superconducting*  $La_{2-x}Ce_xCuO_4$ , npj Quantum Mater. **5**, 1 (2020).
- [105] A. Nag, M. Zhu, M. Bejas, J. Li, H. C. Robarts, H. Yamase, A. N. Petsch, D. Song, H. Eisaki, A. C. Walters, M. García-Fernández, A. Greco, S. M. Hayden, and K.-J. Zhou, Detection of acoustic plasmons in hole-doped lanthanum and bismuth cuprate superconductors using resonant inelastic x-ray scattering, Phys. Rev. Lett. 125, 257002 (2020).
- [106] A. Singh, H. Y. Huang, C. Lane, J. H. Li, J. Okamoto, S. Komiya, R. S. Markiewicz, A. Bansil, T. K. Lee,

A. Fujimori, C. T. Chen, and D. J. Huang, *Acoustic plasmons and conducting carriers in hole-doped cuprate superconductors*, Phys. Rev. B **105**, 235105 (2022).

- [107] M. Hepting, M. Bejas, A. Nag, H. Yamase, N. Coppola, D. Betto, C. Falter, M. Garcia-Fernandez, S. Agrestini, K.-J. Zhou, M. Minola, C. Sacco, L. Maritato, P. Orgiani, H. I. Wei, K. M. Shen, D. G. Schlom, A. Galdi, A. Greco, and B. Keimer, *Gapped collective charge excitations and interlayer hopping in cuprate superconductors*, Phys. Rev. Lett. **129**, 047001 (2022).
- [108] M. Hepting, T. D. Boyko, V. Zimmermann, M. Bejas, Y. E. Suyolcu, P. Puphal, R. J. Green, L. Zinni, J. Kim, D. Casa et al., Evolution of plasmon excitations across the phase diagram of the cuprate superconductor La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub>, Phys. Rev. B **107**, 214516 (2023).
- [109] M. Mitrano, A. A. Husain, S. Vig, A. Kogar, M. S. Rak, S. I. Rubeck, J. Schmalian, B. Uchoa, J. Schneeloch, R. Zhong, G. D. Gu, and P. Abbamonte, *Anomalous density fluctuations in a strange metal*, Proc. Natl. Acad. Sci. U.S.A. **115**, 5392 (2018).
- [110] A. A. Husain, M. Mitrano, M. S. Rak, S. Rubeck, B. Uchoa, K. March, C. Dwyer, J. Schneeloch, R. Zhong, G. D. Gu, and P. Abbamonte, *Crossover of charge fluctuations across the strange metal phase diagram*, Phys. Rev. X 9, 041062 (2019).
- [111] D. Li, K. Lee, B. Y. Wang, M. Osada, S. Crossley, H. R. Lee, Y. Cui, Y. Hikita, and H. Y. Hwang, *Superconductivity in an infinite-layer nickelate*, Nature (London) **572**, 624 (2019).
- [112] D. Li, B. Y. Wang, K. Lee, S. P. Harvey, M. Osada, B. H. Goodge, L. F. Kourkoutis, and H. Y. Hwang, *Superconducting dome in* Nd<sub>1-x</sub>Sr<sub>x</sub>NiO<sub>2</sub> *infinite layer films*, Phys. Rev. Lett. **125**, 027001 (2020).
- [113] G. A. Pan, D. Ferenc Segedin, H. LaBollita, Q. Song, E. M. Nica, B. H. Goodge, A. T. Pierce, S. Doyle, S. Novakov, D. Córdova Carrizales *et al.*, *Superconductivity in a quintuplelayer square-planar nickelate*, Nat. Mater. **21**, 160 (2022).
- [114] H. Sun, M. Huo, X. Hu, J. Li, Z. Liu, Y. Han, L. Tang, Z. Mao, P. Yang, B. Wang, J. Cheng, D.-X. Yao, G.-M. Zhang, and M. Wang, *Signatures of superconductivity near 80 K in a nickelate under high pressure*, Nature (London) **621**, 493 (2023).
- [115] Y. Zhu, D. Peng, E. Zhang, B. Pan, X. Chen, L. Chen, H. Ren, F. Liu, Y. Hao, N. Li *et al.*, *Superconductivity in pressurized trilayer*  $La_4Ni_3O_{10-\delta}$  *single crystals*, Nature (London) **631**, 531 (2024).
- [116] M. Hepting, D. Li, C. J. Jia, H. Lu, E. Paris, Y. Tseng, X. Feng, M. Osada, E. Been, Y. Hikita *et al.*, *Electronic structure of the parent compound of superconducting infinite-layer nickelates*, Nat. Mater. **19**, 381 (2020).
- [117] J. Q. Lin, P. Villar Arribi, G. Fabbris, A. S. Botana, D. Meyers, H. Miao, Y. Shen, D. G. Mazzone, J. Feng, S. G. Chiuzbăian et al., Strong superexchange in a  $d^{9-\delta}$  nickelate revealed by resonant inelastic x-ray scattering, Phys. Rev. Lett. **126**, 087001 (2021).
- [118] H. Lu, M. Rossi, A. Nag, M. Osada, D. F. Li, K. Lee, B. Y. Wang, M. Garcia-Fernandez, S. Agrestini, Z. X. Shen, E. M. Been, B. Moritz, T. P. Devereaux, J. Zaanen, H. Y. Hwang, K.-J. Zhou, and W. S. Lee, *Magnetic excitations in infinite-layer nickelates*, Science **373**, 213 (2021).

- [119] Y. Shen, J. Sears, G. Fabbris, J. Li, J. Pelliciari, I. Jarrige, X. He, I. Božović, M. Mitrano, J. Zhang, J. F. Mitchell, A. S. Botana, V. Bisogni, M. R. Norman, S. Johnston, and M. P. M. Dean, *Role of oxygen states in the low valence nickelate* La<sub>4</sub>Ni<sub>3</sub>O<sub>8</sub>, Phys. Rev. X **12**, 011055 (2022).
- [120] L. Zinni, M. Bejas, H. Yamase, and A. Greco, *Low-energy plasmon excitations in infinite-layer nickelates*, Phys. Rev. B 107, 014503 (2023).
- [121] M. Rossi, M. Osada, J. Choi, S. Agrestini, D. Jost, Y. Lee, H. Lu, B. Y. Wang, K. Lee, A. Nag, Y.-D. Chuang, C.-T. Kuo, S.-J. Lee, B. Moritz, T. P. Devereaux, Z.-X. Shen, J.-S. Lee, K.-J. Zhou, H. Y. Hwang, and W.-S. Lee, A broken translational symmetry state in an infinite-layer nickelate, Nat. Phys. 18, 869 (2022).
- [122] C. C. Tam, J. Choi, X. Ding, S. Agrestini, A. Nag, M. Wu, B. Huang, H. Luo, P. Gao, M. García-Fernández, L. Qiao, and K.-J. Zhou, *Charge density waves in infinite-layer* NdNiO<sub>2</sub> nickelates, Nat. Mater. 21, 1116 (2022).
- [123] G. Krieger, L. Martinelli, S. Zeng, L. E. Chow, K. Kummer, R. Arpaia, M. Moretti Sala, N. B. Brookes, A. Ariando, N. Viart, M. Salluzzo, G. Ghiringhelli, and D. Preziosi, *Charge* and spin order dichotomy in NdNiO<sub>2</sub> driven by the capping layer, Phys. Rev. Lett. **129**, 027002 (2022).
- [124] Y. Shen, J. Sears, G. Fabbris, J. Li, J. Pelliciari, M. Mitrano, W. He, J. Zhang, J. F. Mitchell, V. Bisogni, M. R. Norman, S. Johnston, and M. P. M. Dean, *Electronic character of charge order in square-planar low-valence nickelates*, Phys. Rev. X 13, 011021 (2023).
- [125] C. T. Parzyck, N. K. Gupta, Y. Wu, V. Anil, L. Bhatt, M. Bouliane, R. Gong, B. Z. Gregory, A. Luo, R. Sutarto, F. He, Y. D. Chuang, T. Zhou, G. Herranz, L. F. Kourkoutis, A. Singer, D. G. Schlom, D. G. Hawthorn, and K. M. Shen, *Absence of 3a<sub>0</sub> charge density wave order in the infinite-layer nickelate* NdNiO<sub>2</sub>, Nat. Mater. 23, 486 (2024).
- [126] R. Comin and A. Damascelli, *Resonant x-ray scattering studies of charge order in cuprates*, Annu. Rev. Condens. Matter Phys. 7, 369 (2016).
- [127] A. Frano, S. Blanco-Canosa, B. Keimer, and R.J. Birgeneau, *Charge ordering in superconducting copper* oxides, J. Phys. Condens. Matter **32**, 374005 (2020).
- [128] E. Fradkin, S. A. Kivelson, and J. M. Tranquada, *Colloquium: Theory of intertwined orders in high temperature superconductors*, Rev. Mod. Phys. 87, 457 (2015).
- [129] B.-X. Zheng, C.-M. Chung, P. Corboz, G. Ehlers, M.-P. Qin, R. M. Noack, H. Shi, S. R. White, S. Zhang, and G. K.-L. Chan, *Stripe order in the underdoped region of the twodimensional Hubbard model*, Science **358**, 1155 (2017).
- [130] E. W. Huang, C. B. Mendl, H.-C. Jiang, B. Moritz, and T. P. Devereaux, *Stripe order from the perspective of the Hubbard model*, npj Quantum Mater. 3, 22 (2018).
- [131] M. Qin *et al.* (Simons Collaboration on the Many-Electron Problem), *Absence of superconductivity in the pure twodimensional Hubbard model*, Phys. Rev. X **10**, 031016 (2020).
- [132] P. Mai, S. Karakuzu, G. Balduzzi, S. Johnston, and T. A. Maier, Intertwined spin, charge, and pair correlations in the two-dimensional Hubbard model in the thermodynamic limit, Proc. Natl. Acad. Sci. U.S.A. 119, e2112806119 (2022).

- [133] V. J. Emery, S. A. Kivelson, and O. Zachar, Spin-gap proximity effect mechanism of high-temperature superconductivity, Phys. Rev. B 56, 6120 (1997).
- [134] S. A. Kivelson, E. Fradkin, and V. J. Emery, *Electronic liquid-crystal phases of a doped Mott insulator*, Nature (London) **393**, 550 (1998).
- [135] C. Castellani, C. Di Castro, and M. Grilli, Singular quasiparticle scattering in the proximity of charge instabilities, Phys. Rev. Lett. 75, 4650 (1995).
- [136] J. Fink, E. Schierle, E. Weschke, and J. Geck, *Resonant elastic soft x-ray scattering*, Rep. Prog. Phys. **76**, 056502 (2013).
- [137] H. Miao, R. Fumagalli, M. Rossi, J. Lorenzana, G. Seibold, F. Yakhou-Harris, K. Kummer, N. B. Brookes, G. D. Gu, L. Braicovich, G. Ghiringhelli, and M. P. M. Dean, *Formation of incommensurate charge density waves in cuprates*, Phys. Rev. X 9, 031042 (2019).
- [138] Q. Li, H.-Y. Huang, T. Ren, E. Weschke, L. Ju, C. Zou, S. Zhang, Q. Qiu, J. Liu, S. Ding, A. Singh, O. Prokhnenko, D.-J. Huang, I. Esterlis, Y. Wang, Y. Xie, and Y. Peng, *Prevailing charge order in overdoped* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub> beyond the superconducting dome, Phys. Rev. Lett. **131**, 116002 (2023).
- [139] Y. Y. Peng, A. A. Husain, M. Mitrano, S. X.-L. Sun, T. A. Johnson, A. V. Zakrzewski, G. J. MacDougall, A. Barbour, I. Jarrige, V. Bisogni, and P. Abbamonte, *Enhanced electron-phonon coupling for charge-density-wave formation in*  $La_{1.8-x}Eu_{0.2}Sr_xCuO_{4+\delta}$ , Phys. Rev. Lett. **125**, 097002 (2020).
- [140] W. S. Lee, K.-J. Zhou, M. Hepting, J. Li, A. Nag, A. C. Walters, M. Garcia-Fernandez, H. C. Robarts, M. Hashimoto, H. Lu, B. Nosarzewski, D. Song, H. Eisaki, Z. X. Shen, B. Moritz, J. Zaanen, and T. P. Devereaux, *Spectroscopic fingerprint of charge order melting driven by quantum fluctuations in a cuprate*, Nat. Phys. **17**, 53 (2021).
- [141] H.-H. Kim, S. M. Souliou, M. E. Barber, E. Lefrançois, M. Minola, M. Tortora, R. Heid, N. Nandi, R. A. Borzi, G. Garbarino, A. Bosak, J. Porras, T. Loew, M. König, P. J. W. Moll, A. P. Mackenzie, B. Keimer, C. W. Hicks, and M. L. Tacon, Uniaxial pressure control of competing orders in a high-temperature superconductor, Science 362, 1040 (2018).
- [142] H.-H. Kim, E. Lefrançois, K. Kummer, R. Fumagalli, N. B. Brookes, D. Betto, S. Nakata, M. Tortora, J. Porras, T. Loew, M. E. Barber, L. Braicovich, A. P. Mackenzie, C. W. Hicks, B. Keimer, M. Minola, and M. Le Tacon, *Charge density waves in* YBa<sub>2</sub>Cu<sub>3</sub>O<sub>6.67</sub> *probed by resonant x-ray scattering under uniaxial compression*, Phys. Rev. Lett. **126**, 037002 (2021).
- [143] P. Monthoux, D. Pines, and G. Lonzarich, *Superconductivity without phonons*, Nature (London) **450**, 1177 (2007).
- [144] R. A. Cooper, Y. Wang, B. Vignolle, O. J. Lipscombe, S. M. Hayden, Y. Tanabe, T. Adachi, Y. Koike, M. Nohara, H. Takagi, C. Proust, and N. E. Hussey, *Anomalous criticality in the electrical resistivity of* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub>, Science **323**, 603 (2009).
- [145] B. Michon, C. Girod, S. Badoux, J. Kačmarčík, Q. Ma, M. Dragomir, H. A. Dabkowska, B. D. Gaulin, J.-S. Zhou, S. Pyon, T. Takayama, H. Takagi, S. Verret,

PHYS. REV. X 14, 040501 (2024)

N. Doiron-Leyraud, C. Marcenat, L. Taillefer, and T. Klein, *Thermodynamic signatures of quantum criticality in cuprate superconductors*, Nature (London) **567**, 218 (2019).

- [146] H. Y. Huang, A. Singh, C. Y. Mou, S. Johnston, A. F. Kemper, J. van den Brink, P.J. Chen, T. K. Lee, J. Okamoto, Y. Y. Chu, J. H. Li, S. Komiya, A. C. Komarek, A. Fujimori, C. T. Chen, and D. J. Huang, *Quantum fluctuations of charge order induce phonon softening in a superconducting cuprate*, Phys. Rev. X 11, 041038 (2021).
- [147] R. Arpaia, L. Martinelli, M. M. Sala, S. Caprara, A. Nag, N. B. Brookes, P. Camisa, Q. Li, Q. Gao, X. Zhou, M. Garcia-Fernandez, K.-J. Zhou, E. Schierle, T. Bauch, Y. Y. Peng, C. Di Castro, M. Grilli, F. Lombardi, L. Braicovich, and G. Ghiringhelli, *Signature of quantum criticality in cuprates by charge density fluctuations*, Nat. Commun. 14, 7198 (2023).
- [148] J. Miyawaki, K. Fujii, T. Imazono, K. Horiba, Y. Ohtsubo, N. Inami, T. Nakatani, K. Inaba, A. Agui, H. Kimura, and M. Takahasi, *Design of ultrahigh energy resolution RIXS beamline at NanoTerasu*, J. Phys. Conf. Ser. 2380, 012030 (2022).
- [149] Y. Zhou, K. Kanoda, and T.-K. Ng, *Quantum spin liquid states*, Rev. Mod. Phys. 89, 025003 (2017).
- [150] L. Savary and L. Balents, *Quantum spin liquids: A review*, Rep. Prog. Phys. **80**, 016502 (2017).
- [151] H. Bethe, On the theory of metals: I. Eigenvalues and eigenfunctions of the linear atom chain, J. Phys. 71, 205 (1931).
- [152] J. des Cloizeaux and J. J. Pearson, Spin-wave spectrum of the antiferromagnetic linear chain, Phys. Rev. 128, 2131 (1962).
- [153] A. C. Walters, T. G. Perring, J.-S. Caux, A. T. Savici, G. D. Gu, C.-C. Lee, W. Ku, and I. A. Zaliznyak, *Effect of covalent bonding on magnetism and the missing neutron intensity in copper oxide compounds*, Nat. Phys. 5, 867 (2009).
- [154] J.-S. Caux and R. Hagemans, *The four-spinon dynamical structure factor of the Heisenberg chain*, J. Stat. Mech. (2006), P12013.
- [155] Y. Shen, J. Sears, G. Fabbris, A. Weichselbaum, W. Yin, H. Zhao, D. G. Mazzone, H. Miao, M. H. Upton, D. Casa, R. Acevedo-Esteves, C. Nelson, A. M. Barbour, C. Mazzoli, G. Cao, and M. P. M. Dean, *Emergence of spinons in layered trimer iridate* Ba<sub>4</sub>Ir<sub>3</sub>O<sub>10</sub>, Phys. Rev. Lett. **129**, 207201 (2022).
- [156] J. Schlappa, U. Kumar, K. J. Zhou, S. Singh, M. Mourigal, V. N. Strocov, A. Revcolevschi, L. Patthey, H. M. Rønnow, S. Johnston, and T. Schmitt, *Probing multi-spinon excitations outside of the two-spinon continuum in the antiferromagnetic spin chain cuprate* Sr<sub>2</sub>CuO<sub>3</sub>, Nat. Commun. 9, 5394 (2018).
- [157] A. Kitaev, Anyons in an exactly solved model and beyond, Ann. Phys. (Amsterdam) 321, 2 (2006).
- [158] Z. Nussinov and J. van den Brink, *Compass models: Theory and physical motivations*, Rev. Mod. Phys. 87, 1 (2015).
- [159] S. M. Winter, A. A. Tsirlin, M. Daghofer, J. van den Brink, Y. Singh, P. Gegenwart, and R. Valentí, *Models and*

*materials for generalized Kitaev magnetism*, J. Phys. Condens. Matter **29**, 493002 (2017).

- [160] S. Trebst and C. Hickey, *Kitaev materials*, Phys. Rep. 950, 1 (2022).
- [161] Y. Motome and J. Nasu, *Hunting Majorana fermions* in Kitaev magnets, J. Phys. Soc. Jpn. 89, 012002 (2020).
- [162] H. Takagi, T. Takayama, G. Jackeli, G. Khaliullin, and S. E. Nagler, *Concept and realization of Kitaev quantum spin liquids*, Nat. Rev. Phys. 1, 264 (2019).
- [163] S. Kim, B. Yuan, and Y.-J. Kim, α-RuCl<sub>3</sub> and other Kitaev materials, APL Mater. 10, 080903 (2022).
- [164] A. Banerjee, C. A. Bridges, J.-Q. Yan, A. A. Aczel, L. Li, M. B. Stone, G. E. Granroth, M. D. Lumsden, Y. Yiu, J. Knolle, S. Bhattacharjee, D. L. Kovrizhin, R. Moessner, D. A. Tennant, D. G. Mandrus, and S. E. Nagler, *Proximate Kitaev quantum spin liquid behaviour in a honeycomb magnet*, Nat. Mater. 15, 733 (2016).
- [165] A. Banerjee, J. Yan, J. Knolle, C. A. Bridges, M. B. Stone, M. D. Lumsden, D. G. Mandrus, D. A. Tennant, R. Moessner, and S. E. Nagler, *Neutron scattering in the proximate quantum spin liquid* α-RuCl<sub>3</sub>, Science 356, 1055 (2017).
- [166] A. de la Torre, B. Zager, F. Bahrami, M. DiScala, J. R. Chamorro, M. H. Upton, G. Fabbris, D. Haskel, D. Casa, T. M. McQueen, F. Tafti, and K. W. Plumb, *Enhanced hybridization in the electronic ground state of the intercalated honeycomb iridate* Ag<sub>3</sub>LiIr<sub>2</sub>O<sub>6</sub>, Phys. Rev. B 104, L100416 (2021).
- [167] B. W. Lebert, S. Kim, V. Bisogni, I. Jarrige, A. M. Barbour, and Y.-J. Kim, *Resonant inelastic x-ray scattering study of* α-RuCl<sub>3</sub>: A progress report, J. Phys. Condens. Matter 32, 144001 (2020).
- [168] H. Suzuki, H. Liu, J. Bertinshaw, K. Ueda, H. Kim, S. Laha, D. Weber, Z. Yang, L. Wang, H. Takahashi, K. Fürsich, M. Minola, B. V. Lotsch, B. J. Kim, H. Yavaş, M. Daghofer, J. Chaloupka, G. Khaliullin, H. Gretarsson, and B. Keimer, *Proximate ferromagnetic state in the Kitaev model material* α-RuCl<sub>3</sub>, Nat. Commun. 12, 4512 (2021).
- [169] Y. Shen, Y.-D. Li, H. Wo, Y. Li, S. Shen, B. Pan, Q. Wang, H. C. Walker, P. Steffens, M. Boehm, Y. Hao, D. L. Quintero-Castro, L. W. Harriger, M. D. Frontzek, L. Hao, S. Meng, Q. Zhang, G. Chen, and J. Zhao, *Evidence* for a spinon Fermi surface in a triangular-lattice quantum-spin-liquid candidate, Nature (London) 540, 559 (2016).
- [170] J. A. M. Paddison, M. Daum, Z. Dun, G. Ehlers, Y. Liu, M. B. Stone, H. Zhou, and M. Mourigal, *Continuous excitations of the triangular-lattice quantum spin liquid* YbMgGaO<sub>4</sub>, Nat. Phys. **13**, 117 (2017).
- [171] Z. Ma, J. Wang, Z.-Y. Dong, J. Zhang, S. Li, S.-H. Zheng, Y. Yu, W. Wang, L. Che, K. Ran, S. Bao, Z. Cai, P. Čermák, A. Schneidewind, S. Yano, J. S. Gardner, X. Lu, S.-L. Yu, J.-M. Liu, S. Li, J.-X. Li, and J. Wen, *Spin-glass* ground state in a triangular-lattice compound YbZnGaO<sub>4</sub>, Phys. Rev. Lett. **120**, 087201 (2018).
- [172] U. Kumar, A. Nag, J. Li, H. C. Robarts, A. C. Walters, M. García-Fernández, R. Saint-Martin, A. Revcolevschi, J. Schlappa, T. Schmitt, S. Johnston, and K.-J. Zhou, Unraveling higher-order contributions to spin excitations

probed using resonant inelastic x-ray scattering, Phys. Rev. B **106**, L060406 (2022).

- [173] M. W. Haverkort, Theory of resonant inelastic x-ray scattering by collective magnetic excitations, Phys. Rev. Lett. 105, 167404 (2010).
- [174] A. Nag, H. C. Robarts, F. Wenzel, J. Li, H. Elnaggar, R.-P. Wang, A. C. Walters, M. García-Fernández, F. M. F. de Groot, M. W. Haverkort, and K.-J. Zhou, *Many-body physics of single and double spin-flip excitations in* NiO, Phys. Rev. Lett. **124**, 067202 (2020).
- [175] A. Nag, A. Nocera, S. Agrestini, M. Garcia-Fernandez, A. Walters, S.-W. Cheong, S. Johnston, and K.-J. Zhou, *Quadrupolar magnetic excitations in an isotropic spin-1 antiferromagnet*, Nat. Commun. 13, 1 (2022).
- [176] G. B. Halász, S. Kourtis, J. Knolle, and N. B. Perkins, Observing spin fractionalization in the Kitaev spin liquid via temperature evolution of indirect resonant inelastic x-ray scattering, Phys. Rev. B 99, 184417 (2019).
- [177] G. B. Halász, B. Perreault, and N. B. Perkins, Probing spinon nodal structures in three-dimensional Kitaev spin liquids, Phys. Rev. Lett. 119, 097202 (2017).
- [178] W. M. H. Natori and J. Knolle, Dynamics of a twodimensional quantum spin-orbital liquid: Spectroscopic signatures of fermionic magnons, Phys. Rev. Lett. 125, 067201 (2020).
- [179] W. M. H. Natori, M. Daghofer, and R. G. Pereira, *Dynamics of a*  $j = \frac{3}{2}$  quantum spin liquid, Phys. Rev. B **96**, 125109 (2017).
- [180] W.-H. Ko, Z.-X. Liu, T.-K. Ng, and P. A. Lee, *Raman* signature of the U(1) Dirac spin-liquid state in the spin- $\frac{1}{2}$  kagome system, Phys. Rev. B **81**, 024414 (2010).
- [181] I. Rousochatzakis, S. Kourtis, J. Knolle, R. Moessner, and N. B. Perkins, *Quantum spin liquid at finite temperature: Proximate dynamics and persistent typicality*, Phys. Rev. B 100, 045117 (2019).
- [182] Y. Fu, M.-L. Lin, L. Wang, Q. Liu, L. Huang, W. Jiang, Z. Hao, C. Liu, H. Zhang, X. Shi, J. Zhang, J. Dai, D. Yu, F. Ye, P. A. Lee, P.-H. Tan, and J.-W. Mei, *Dynamic finger-print of fractionalized excitations in single-crystalline* Cu<sub>3</sub>Zn(OH)<sub>6</sub>FBr, Nat. Commun. **12**, 3048 (2021).
- [183] L. Martinelli, D. Betto, K. Kummer, R. Arpaia, L. Braicovich, D. Di Castro, N.B. Brookes, M. Moretti Sala, and G. Ghiringhelli, *Fractional spin excitations in the infinite-layer cuprate* CaCuO<sub>2</sub>, Phys. Rev. X 12, 021041 (2022).
- [184] L. Braicovich, M. Minola, G. Dellea, M. Le Tacon, M. Moretti Sala, C. Morawe, J.-C. Peffen, R. Supruangnet, F. Yakhou, G. Ghiringhelli, and N. B. Brookes, *The simultaneous measurement of energy and linear polarization of the scattered radiation in resonant inelastic soft x-ray scattering*, Rev. Sci. Instrum. **85**, 115104 (2014).
- [185] J. Kim, X. Shi, D. Casa, J. Qian, X. Huang, and T. Gog, Collimating Montel mirror as part of a multi-crystal analyzer system for resonant inelastic x-ray scattering, J. Synchrotron Radiat. 23, 880 (2016).
- [186] Č. Brukner, V. Vedral, and A. Zeilinger, *Crucial role of quantum entanglement in bulk properties of solids*, Phys. Rev. A 73, 012110 (2006).
- [187] V. Coffman, J. Kundu, and W. K. Wootters, *Distributed entanglement*, Phys. Rev. A 61, 052306 (2000).

- [188] P. Hyllus, W. Laskowski, R. Krischek, C. Schwemmer, W. Wieczorek, H. Weinfurter, L. Pezzé, and A. Smerzi, *Fisher information and multiparticle entanglement*, Phys. Rev. A 85, 022321 (2012).
- [189] L. Amico, F. Baroni, A. Fubini, D. Patanè, V. Tognetti, and P. Verrucchi, *Divergence of the entanglement range in lowdimensional quantum systems*, Phys. Rev. A 74, 022322 (2006).
- [190] L. Amico, R. Fazio, A. Osterloh, and V. Vedral, *Entangle-ment in many-body systems*, Rev. Mod. Phys. **80**, 517 (2008).
- [191] P. Hauke, M. Heyl, L. Tagliacozzo, and P. Zoller, *Measuring multipartite entanglement through dynamic susceptibilities*, Nat. Phys. **12**, 778 (2016).
- [192] G. Mathew, S. L. L. Silva, A. Jain, A. Mohan, D. T. Adroja, V. G. Sakai, C. V. Tomy, A. Banerjee, R. Goreti, Ashwathi V. N., R. Singh, and D. Jaiswal-Nagar, *Experimental realization of multipartite entanglement via quantum Fisher information in a uniform antiferromagnetic quantum spin chain*, Phys. Rev. Res. 2, 043329 (2020).
- [193] A. Scheie, P. Laurell, A. M. Samarakoon, B. Lake, S. E. Nagler, G. E. Granroth, S. Okamoto, G. Alvarez, and D. A. Tennant, *Witnessing entanglement in quantum magnets using neutron scattering*, Phys. Rev. B 103, 224434 (2021).
- [194] P. Laurell, A. Scheie, C. J. Mukherjee, M. M. Koza, M. Enderle, Z. Tylczynski, S. Okamoto, R. Coldea, D. A. Tennant, and G. Alvarez, *Quantifying and controlling entanglement in the quantum magnet* Cs<sub>2</sub>CoCl<sub>4</sub>, Phys. Rev. Lett. **127**, 037201 (2021).
- [195] A. O. Scheie, E. A. Ghioldi, J. Xing, J. A. M. Paddison, N. E. Sherman, M. Dupont, L. D. Sanjeewa, S. Lee, A. J. Woods, D. Abernathy *et al.*, *Proximate spin liquid and fractionalization in the triangular antiferromagnet* KYbSe<sub>2</sub>, Nat. Phys. **20**, 74 (2024).
- [196] V. Menon, N. E. Sherman, M. Dupont, A. O. Scheie, D. A. Tennant, and J. E. Moore, *Multipartite entanglement in the one-dimensional spin-*<sup>1</sup>/<sub>2</sub> *Heisenberg antiferromagnet*, Phys. Rev. B **107**, 054422 (2023).
- [197] C. Jia, K. Wohlfeld, Y. Wang, B. Moritz, and T. P. Devereaux, Using RIXS to uncover elementary charge and spin excitations, Phys. Rev. X 6, 021020 (2016).
- [198] T. Ren, Y. Shen, S. F. R. TenHuisen, J. Sears, W. He, M. H. Upton, D. Casa, P. Becker, M. Mitrano, M. P. M. Dean, and R. M. Konik, *Witnessing quantum entanglement using resonant inelastic x-ray scattering*, arXiv:2404.05850.
- [199] B. Yang, Y. M. Goh, S. H. Sung, G. Ye, S. Biswas, D. A. S. Kaib, R. Dhakal, S. Yan, C. Li, S. Jiang, F. Chen, H. Lei, R. He, R. Valentí, S. M. Winter, R. Hovden, and A. W. Tsen, *Magnetic anisotropy reversal driven by structural symmetry-breaking in monolayer* α-RuCl<sub>3</sub>, Nat. Mater. **22**, 50 (2023).
- [200] J. H. Gruenewald, J. Kim, H. S. Kim, J. M. Johnson, J. Hwang, M. Souri, J. Terzic, S. H. Chang, A. Said, J. W. Brill, G. Cao, H.-Y. Kee, and S. S. A. Seo, *Engineering 1D* quantum stripes from superlattices of 2D layered materials, Adv. Mater. 29, 1603798 (2017).
- [201] E. Lesne, Y. G. Sağlam, R. Battilomo, M. T. Mercaldo, T. C. van Thiel, U. Filippozzi, C. Noce, M. Cuoco, G. A. Steele, C. Ortix, and A. D. Caviglia, *Designing spin and*

*orbital sources of Berry curvature at oxide interfaces*, Nat. Mater. **22**, 576 (2023).

- [202] J. Röchner, L. Balents, and K. P. Schmidt, *Spin liquid and quantum phase transition without symmetry breaking in a frustrated three-dimensional Ising model*, Phys. Rev. B 94, 201111(R) (2016).
- [203] Y. H. Wang, H. Steinberg, P. Jarillo-Herrero, and N. Gedik, Observation of Floquet-Bloch states on the surface of a topological insulator, Science 342, 453 (2013).
- [204] J. W. McIver, B. Schulte, F. U. Stein, T. Matsuyama, G. Jotzu, G. Meier, and A. Cavalleri, *Light-induced anomalous Hall effect in graphene*, Nat. Phys. **16**, 38 (2019).
- [205] S. Zhou, C. Bao, B. Fan, H. Zhou, Q. Gao, H. Zhong, T. Lin, H. Liu, P. Yu, P. Tang, S. Meng, W. Duan, and S. Zhou, *Pseudospin-selective Floquet band engineering in black phosphorus*, Nature (London) **614**, 75 (2023).
- [206] S. Ito, M. Schüler, M. Meierhofer, S. Schlauderer, J. Freudenstein, J. Reimann, D. Afanasiev, K. A. Kokh, O. E. Tereshchenko, J. Güdde, M. A. Sentef, U. Höfer, and R. Huber, *Build-up and dephasing of Floquet–Bloch bands on subcycle timescales*, Nature (London) **616**, 696 (2023).
- [207] A. S. Disa, M. Fechner, T. F. Nova, B. Liu, M. Först, D. Prabhakaran, P. G. Radaelli, and A. Cavalleri, *Polarizing* an antiferromagnet by optical engineering of the crystal field, Nat. Phys. 16, 937 (2020).
- [208] D. Afanasiev, J. Hortensius, B. Ivanov, A. Sasani, E. Bousquet, Y. Blanter, R. Mikhaylovskiy, A. Kimel, and A. Caviglia, *Ultrafast control of magnetic interactions via light-driven phonons*, Nat. Mater. 20, 607 (2021).
- [209] A. S. Disa, J. Curtis, M. Fechner, A. Liu, A. von Hoegen, M. Först, T. F. Nova, P. Narang, A. Maljuk, A. V. Boris, B. Keimer, and A. Cavalleri, *Photo-induced high-temperature ferromagnetism in* YTiO<sub>3</sub>, Nature (London) **617**, 73 (2023).
- [210] T. Kubacka, J. A. Johnson, M. C. Hoffmann, C. Vicario, S. de Jong, P. Beaud, S. Grübel, S.-W. Huang, L. Huber, L. Patthey *et al.*, *Large-amplitude spin dynamics driven by a THz pulse in resonance with an electromagnon*, Science 343, 1333 (2014).
- [211] T. F. Nova, A. S. Disa, M. Fechner, and A. Cavalleri, *Metastable ferroelectricity in optically strained* SrTiO<sub>3</sub>, Science 364, 1075 (2019).
- [212] X. Li, T. Qiu, J. Zhang, E. Baldini, J. Lu, A. M. Rappe, and K. A. Nelson, *Terahertz field-induced ferroelectricity in quantum paraelectric* SrTiO<sub>3</sub>, Science **364**, 1079 (2019).
- [213] E. J. Sie, C. M. Nyby, C. D. Pemmaraju, S. J. Park, X. Shen, J. Yang, M. C. Hoffmann, B. K. Ofori-Okai, R. Li, A. H. Reid *et al.*, *An ultrafast symmetry switch in a Weyl semimetal*, Nature (London) **565**, 61 (2019).
- [214] D. Fausti, R. I. Tobey, N. Dean, S. Kaiser, A. Dienst, M. C. Hoffmann, S. Pyon, T. Takayama, H. Takagi, and A. Cavalleri, *Light-induced superconductivity in a stripeordered cuprate*, Science **331**, 189 (2011).
- [215] W. Hu, S. Kaiser, D. Nicoletti, C. R. Hunt, I. Gierz, M. C. Hoffmann, M. Le Tacon, T. Loew, B. Keimer, and A. Cavalleri, *Optically enhanced coherent transport in* YBa<sub>2</sub>Cu<sub>3</sub>O<sub>6.5</sub> by ultrafast redistribution of interlayer coupling, Nat. Mater. 13, 705 (2014).

- [216] M. Mitrano, A. Cantaluppi, D. Nicoletti, S. Kaiser, A. Perucchi, S. Lupi, P. Di Pietro, D. Pontiroli, M. Riccò, S. R. Clark, D. Jaksch, and A. Cavalleri, *Possible light-induced superconductivity in* K<sub>3</sub>C<sub>60</sub> at high temperature, Nature (London) **530**, 461 (2016).
- [217] C. Giannetti, M. Capone, D. Fausti, M. Fabrizio, F. Parmigiani, and D. Mihailovic, Ultrafast optical spectroscopy of strongly correlated materials and high-temperature superconductors: A non-equilibrium approach, Adv. Phys. 65, 58 (2016).
- [218] J. Zhang and R. Averitt, *Dynamics and control in complex transition metal oxides*, Annu. Rev. Mater. Res. 44, 19 (2014).
- [219] S. P. Weathersby, G. Brown, M. Centurion, T. F. Chase, R. Coffee, J. Corbett, J. P. Eichner, J. C. Frisch, A. R. Fry, M. Gühr et al., Mega-electron-volt ultrafast electron diffraction at SLAC National Accelerator Laboratory, Rev. Sci. Instrum. 86, 073702 (2015).
- [220] D. Filippetto, P. Musumeci, R. K. Li, B. J. Siwick, M. R. Otto, M. Centurion, and J. P. F. Nunes, *Ultrafast electron diffraction: Visualizing dynamic states of matter*, Rev. Mod. Phys. **94**, 045004 (2022).
- [221] U. Bovensiepen and P. Kirchmann, *Elementary relaxation processes investigated by femtosecond photoelectron spectroscopy of two-dimensional materials*, Laser Photonics Rev. 6, 589 (2012).
- [222] F. Boschini, M. Zonno, and A. Damascelli, *Time-resolved ARPES studies of quantum materials*, Rev. Mod. Phys. 96, 015003 (2024).
- [223] M. Mitrano and Y. Wang, Probing light-driven quantum materials with ultrafast resonant inelastic x-ray scattering, Commun. Phys. 3, 184 (2020).
- [224] Y. Wang, Y. Chen, T. P. Devereaux, B. Moritz, and M. Mitrano, X-ray scattering from light-driven spin fluctuations in a doped Mott insulator, Commun. Phys. 4, 212 (2021).
- [225] Y. Murakami, S. Takayoshi, T. Kaneko, Z. Sun, D. Golež, A. J. Millis, and P. Werner, *Exploring nonequilibrium phases of photo-doped Mott insulators with generalized Gibbs ensembles*, Commun. Phys. 5, 23 (2022).
- [226] Y. Murakami, D. Golež, M. Eckstein, and P. Werner, *Photo-induced nonequilibrium states in Mott insulators*, arXiv:2310.05201.
- [227] P. Emma, R. Akre, J. Arthur, R. Bionta, C. Bostedt, J. Bozek, A. Brachmann, P. Bucksbaum, R. Coffee, F.-J. Decker et al., First lasing and operation of an Ångstromwavelength free-electron laser, Nat. Photonics 4, 641 (2010).
- [228] B. W. J. McNeil and N. R. Thompson, *X-ray free-electron lasers*, Nat. Photonics **4**, 814 (2010).
- [229] C. Bostedt, S. Boutet, D. M. Fritz, Z. Huang, H. J. Lee, H. T. Lemke, A. Robert, W. F. Schlotter, J. J. Turner, and G. J. Williams, *Linac Coherent Light Source: The first five years*, Rev. Mod. Phys. 88, 015007 (2016).
- [230] Y. Cao, D. G. Mazzone, D. Meyers, J. P. Hill, X. Liu, S. Wall, and M. P. M. Dean, Ultrafast dynamics of spin and orbital correlations in quantum materials: An energy- and momentum-resolved perspective, Phil. Trans. R. Soc. A 377, 20170480 (2019).
- [231] D. G. Mazzone, D. Meyers, Y. Cao, J. G. Vale, C. D. Dashwood, Y. Shi, A. J. A. James, N. J. Robinson, J. Lin,

V. Thampy *et al.*, *Laser-induced transient magnons in*  $Sr_3Ir_2O_7$  *throughout the Brillouin zone*, Proc. Natl. Acad. Sci. U.S.A. **118**, e2103696118 (2021).

- [232] E. Paris, C. W. Nicholson, S. Johnston, Y. Tseng, M. Rumo, G. Coslovich, S. Zohar, M. F. Lin, V. N. Strocov, R. Saint-Martin, A. Revcolevschi, A. Kemper, W. Schlotter, G. L. Dakovski, C. Monney, and T. Schmitt, *Probing the interplay between lattice dynamics and short-range magnetic correlations in* CuGeO<sub>3</sub> with femtosecond RIXS, npj Quantum Mater. 6, 51 (2021).
- [233] M. Mitrano, S. Lee, A. A. Husain, M. Zhu, G. P. Munoz, Stella X.-L. Sun, Y. I. Joe, A. H. Reid, S. F. Wandel, G. Coslovich, W. Schlotter, T. van Driel, J. Schneeloch, G. D. Gu, N. Goldenfeld, and P. Abbamonte, *Evidence* for photoinduced sliding of the charge-order condensate in La<sub>1.875</sub>Ba<sub>0.125</sub>CuO<sub>4</sub>, Phys. Rev. B **100**, 205125 (2019).
- [234] S. Parchenko, E. Paris, D. McNally, E. Abreu, M. Dantz, E. M. Bothschafter, A. H. Reid, W. F. Schlotter, M.-F. Lin, S. F. Wandel *et al.*, *Orbital dynamics during an ultrafast insulator to metal transition*, Phys. Rev. Res. 2, 023110 (2020).
- [235] T. Suzuki, T. Someya, T. Hashimoto, S. Michimae, M. Watanabe, M. Fujisawa, T. Kanai, N. Ishii, J. Itatani, S. Kasahara, Y. Matsuda, T. Shibauchi, K. Okazaki, and S. Shin, *Photoinduced possible superconducting state with long-lived disproportionate band filling in* FeSe, Commun. Phys. 2, 115 (2019).
- [236] S. Fava, G. De Vecchi, G. Jotzu, M. Buzzi, T. Gebert, Y. Liu, B. Keimer, and A. Cavalleri, *Magnetic field expulsion in optically driven* YBa<sub>2</sub>Cu<sub>3</sub>O<sub>6.48</sub>, Nature (London) 632, 75 (2024).
- [237] M. Buzzi, D. Nicoletti, M. Fechner, N. Tancogne-Dejean, M. A. Sentef, A. Georges, T. Biesner, E. Uykur, M. Dressel, A. Henderson *et al.*, *Photomolecular high-temperature superconductivity*, Phys. Rev. X **10**, 031028 (2020).
- [238] M. Budden, T. Gebert, M. Buzzi, G. Jotzu, E. Wang, T. Matsuyama, G. Meier, Y. Laplace, D. Pontiroli, M. Riccò, F. Schlawin, D. Jaksch, and A. Cavalleri, *Evidence* for metastable photo-induced superconductivity in K<sub>3</sub>C<sub>60</sub>, Nat. Phys. **17**, 611 (2021).
- [239] Z. M. Raines, V. Stanev, and V. M. Galitski, *Enhancement of superconductivity via periodic modulation in a three-dimensional model of cuprates*, Phys. Rev. B **91**, 184506 (2015).
- [240] M. Babadi, M. Knap, I. Martin, G. Refael, and E. Demler, *Theory of parametrically amplified electronphonon superconductivity*, Phys. Rev. B **96**, 014512 (2017).
- [241] S. J. Denny, S. R. Clark, Y. Laplace, A. Cavalleri, and D. Jaksch, *Proposed parametric cooling of bilayer cuprate superconductors by terahertz excitation*, Phys. Rev. Lett. 114, 137001 (2015).
- [242] A. Nava, C. Giannetti, A. Georges, E. Tosatti, and M. Fabrizio, *Cooling quasiparticles in* A<sub>3</sub>C<sub>60</sub> *fullerides by excitonic mid-infrared absorption*, Nat. Phys. 14, 154 (2018).
- [243] D. M. Kennes, E. Y. Wilner, D. R. Reichman, and A. J. Millis, *Transient superconductivity from electronic*

squeezing of optically pumped phonons, Nat. Phys. 13, 479 (2017).

- [244] S. Kaiser, C. R. Hunt, D. Nicoletti, W. Hu, I. Gierz, H. Y. Liu, M. Le Tacon, T. Loew, D. Haug, B. Keimer, and A. Cavalleri, *Optically induced coherent transport far above*  $T_c$  in underdoped YBa<sub>2</sub>Cu<sub>3</sub>O<sub>6+ $\delta$ </sub>, Phys. Rev. B **89**, 184516 (2014).
- [245] B. Liu, M. Först, M. Fechner, D. Nicoletti, J. Porras, T. Loew, B. Keimer, and A. Cavalleri, *Pump frequency resonances for light-induced incipient superconductivity in* YBa<sub>2</sub>Cu<sub>3</sub>O<sub>6.5</sub>, Phys. Rev. X 10, 011053 (2020).
- [246] D. Nicoletti, E. Casandruc, Y. Laplace, V. Khanna, C. R. Hunt, S. Kaiser, S. S. Dhesi, G. D. Gu, J. P. Hill, and A. Cavalleri, *Optically induced superconductivity in striped*  $La_{2-x}Ba_xCuO_4$  *by polarization-selective excitation in the near infrared*, Phys. Rev. B **90**, 100503(R) (2014).
- [247] D. J. Scalapino, A common thread: The pairing interaction for unconventional superconductors, Rev. Mod. Phys. 84, 1383 (2012).
- [248] M. Först, C. Manzoni, S. Kaiser, Y. Tomioka, Y. Tokura, R. Merlin, and A. Cavalleri, *Nonlinear phononics as an ultrafast route to lattice control*, Nat. Phys. 7, 854 (2011).
- [249] A. Subedi, A. Cavalleri, and A. Georges, *Theory of nonlinear phononics for coherent light control of solids*, Phys. Rev. B 89, 220301(R) (2014).
- [250] A. S. Disa, T. F. Nova, and A. Cavalleri, *Engineering crystal structures with light*, Nat. Phys. 17, 1087 (2021).
- [251] D. R. Baykusheva, H. Jang, A. A. Husain, S. Lee, S. F. R. TenHuisen, P. Zhou, S. Park, H. Kim, J.-K. Kim, H.-D. Kim, M. Kim, S.-Y. Park, P. Abbamonte, B. J. Kim, G. D. Gu, Y. Wang, and M. Mitrano, *Ultrafast renormalization* of the on-site Coulomb repulsion in a cuprate superconductor, Phys. Rev. X 12, 011013 (2022).
- [252] K. Isoyama, N. Yoshikawa, K. Katsumi, J. Wong, N. Shikama, Y. Sakishita, F. Nabeshima, A. Maeda, and R. Shimano, *Light-induced enhancement of superconductiv-ity in iron-based superconductor* FeSe<sub>0.5</sub>Te<sub>0.5</sub>, Commun. Phys. 4, 160 (2021).
- [253] S. Gerber, S.-L. Yang, D. Zhu, H. Soifer, J. A. Sobota, S. Rebec, J. J. Lee, T. Jia, B. Moritz, C. Jia *et al.*, *Femto-second electron-phonon lock-in by photoemission and x-ray free-electron laser*, Science **357**, 71 (2017).
- [254] J. H. Mentink and M. Eckstein, Ultrafast quenching of the exchange interaction in a Mott insulator, Phys. Rev. Lett. 113, 057201 (2014).
- [255] J. H. Mentink, K. Balzer, and M. Eckstein, Ultrafast and reversible control of the exchange interaction in Mott insulators, Nat. Commun. 6, 6708 (2015).
- [256] S. Chaudhary, D. Hsieh, and G. Refael, Orbital Floquet engineering of exchange interactions in magnetic materials, Phys. Rev. B 100, 220403(R) (2019).
- [257] N. Walldorf, D. M. Kennes, J. Paaske, and A. J. Millis, *The antiferromagnetic phase of the Floquet-driven Hubbard model*, Phys. Rev. B **100**, 121110(R) (2019).
- [258] C. N. Yang, η pairing and off-diagonal long-range order in a Hubbard model, Phys. Rev. Lett. 63, 2144 (1989).
- [259] S. Zhang, Pseudospin symmetry and new collective modes of the Hubbard model, Phys. Rev. Lett. 65, 120 (1990).

- [260] T. Kaneko, T. Shirakawa, S. Sorella, and S. Yunoki, *Photoinduced η pairing in the Hubbard model*, Phys. Rev. Lett. **122**, 077002 (2019).
- [261] Y. Murakami, S. Takayoshi, T. Kaneko, A. M. Läuchli, and P. Werner, Spin, charge, and η-spin separation in onedimensional photodoped Mott insulators, Phys. Rev. Lett. 130, 106501 (2023).
- [262] F. Peronaci, O. Parcollet, and M. Schiró, Enhancement of local pairing correlations in periodically driven Mott insulators, Phys. Rev. B 101, 161101(R) (2020).
- [263] H. Suzuki, M. Minola, Y. Lu, Y. Peng, R. Fumagalli, E. Lefrançois, T. Loew, J. Porras, K. Kummer, D. Betto, S. Ishida, H. Eisaki, C. Hu, X. Zhou, M. W. Haverkort, N. B. Brookes, L. Braicovich, G. Ghiringhelli, M. Le Tacon, and B. Keimer, *Probing the energy gap of high-temperature cuprate superconductors by resonant inelastic x-ray scattering*, npj Quantum Mater. **3**, 65 (2018).
- [264] L. Stojchevska, I. Vaskivskyi, T. Mertelj, P. Kusar, D. Svetin, S. Brazovskii, and D. Mihailovic, *Ultrafast switching to a stable hidden quantum state in an electronic crystal*, Science **344**, 177 (2014).
- [265] J. Li, L. Xu, M. Garcia-Fernandez, A. Nag, H. C. Robarts, A. C. Walters, X. Liu, J. Zhou, K. Wohlfeld, J. van den Brink, H. Ding, and K.-J. Zhou, *Unraveling the orbital physics in a canonical orbital system* KCuF<sub>3</sub>, Phys. Rev. Lett. **126**, 106401 (2021).
- [266] J. Li, H. U. R. Strand, P. Werner, and M. Eckstein, *Theory of photoinduced ultrafast switching to a spin-orbital ordered hidden phase*, Nat. Commun. 9, 4581 (2018).
- [267] X. Jia, A. Haldar, J. Kim, Y. Wang, G. Fabbris, K. Ludwig, S. Kourtis, M. Upton, Y. Liu, W. Lu, X. Luo, Y.-P. Sun, D. Casa, S. Sharifzadeh, P. T. Darancet, and Y. Cao, *Interplay of broken symmetry and delocalized excitations in the insulating state of* 1*T*-Tas<sub>2</sub>, Phys. Rev. B **108**, 205105 (2023).
- [268] Y. Tokura, M. Kawasaki, and N. Nagaosa, *Emergent functions of quantum materials*, Nat. Phys. 13, 1056 (2017).
- [269] D. Meyers, K. Nakatsukasa, S. Mu, L. Hao, J. Yang, Y. Cao, G. Fabbris, H. Miao, J. Pelliciari, D. McNally et al., Decoupling carrier concentration and electron-phonon coupling in oxide heterostructures observed with resonant inelastic x-ray scattering, Phys. Rev. Lett. 121, 236802 (2018).
- [270] V. Bisogni, S. Catalano, R. J. Green, M. Gibert, R. Scherwitzl, Y. Huang, V. N. Strocov, P. Zubko, S. Balandeh, J.-M. Triscone, G. Sawatzky, and T. Schmitt, Ground-state oxygen holes and the metal-insulator transition in the negative charge-transfer rare-earth nickel-ates, Nat. Commun. 7, 1 (2016).
- [271] G. Fabbris, D. Meyers, J. Okamoto, J. Pelliciari, A. S. Disa, Y. Huang, Z.-Y. Chen, W. B. Wu, C. T. Chen, S. Ismail-Beigi, C. H. Ahn, F. J. Walker, D. J. Huang, T. Schmitt, and M. P. M. Dean, *Orbital engineering in nickel-ate heterostructures driven by anisotropic oxygen hybridization rather than orbital energy levels*, Phys. Rev. Lett. 117, 147401 (2016).
- [272] K.-J. Zhou, M. Radovic, J. Schlappa, V. Strocov, R. Frison, J. Mesot, L. Patthey, and T. Schmitt, *Localized and delocalized Ti 3d carriers in* LaAlO<sub>3</sub>/SrTiO<sub>3</sub> superlattices

*revealed by resonant inelastic x-ray scattering*, Phys. Rev. B **83**, 201402(R) (2011).

- [273] C. Gong, L. Li, Z. Li, H. Ji, A. Stern, Y. Xia, T. Cao, W. Bao, C. Wang, Y. Wang, Z. Q. Qiu, R. J. Cava, S. G. Louie, J. Xia, and X. Zhang, *Discovery of intrinsic ferromagnetism in two-dimensional van der Waals crystals*, Nature (London) **546**, 265 (2017).
- [274] Q. Song, C. A. Occhialini, E. Ergeçen, B. Ilyas, D. Amoroso, P. Barone, J. Kapeghian, K. Watanabe, T. Taniguchi, A. S. Botana, S. Picozzi, N. Gedik, and R. Comin, *Evidence for a single-layer van der Waals multi-ferroic*, Nature (London) **602**, 601 (2022).
- [275] S. Kang, K. Kim, B. H. Kim, J. Kim, K. I. Sim, J.-U. Lee, S. Lee, K. Park, S. Yun, T. Kim, A. Nag, A. Walters, M. Garcia-Fernandez, J. Li, L. Chapon, K.-J. Zhou, Y.-W. Son, J. H. Kim, H. Cheong, and J.-G. Park, *Coherent many-body exciton in van der Waals antiferromagnet* NiPS<sub>3</sub>, Nature (London) 583, 785 (2020).
- [276] M. F. DiScala, D. Staros, A. de la Torre, A. Lopez, D. Wong, C. Schulz, M. Barkowiak, V. Bisogni, J. Pelliciari, B. Rubenstein, and K. W. Plumb, *Elucidating the role of dimensionality on the electronic structure of the van der Waals antiferromagnet* NiPS<sub>3</sub>, Adv. Phys. Res. **3**, 2300096 (2024).
- [277] One such beam line is the ARI beam line at the National Synchrotron Light Source II, https://www.bnl.gov/nsls2/ beamlines/beamline.php?r=29-ID-2.
- [278] Y. Qi, M. A. Sadi, D. Hu, M. Zheng, Z. Wu, Y. Jiang, and Y. P. Chen, *Recent progress in strain engineering on van der Waals 2D materials: Tunable electrical, electrochemical, magnetic, and optical properties, Adv. Mater.* 35, 2205714 (2023).
- [279] G. Wang, A. Chernikov, M. M. Glazov, T. F. Heinz, X. Marie, T. Amand, and B. Urbaszek, *Colloquium: Excitons in atomically thin transition metal dichalcogenides*, Rev. Mod. Phys. **90**, 021001 (2018).
- [280] J. Frenkel, On the transformation of light into heat in solids. I, Phys. Rev. 37, 17 (1931).
- [281] P. Yu and M. Cardona, Fundamentals of Semiconductors: Physics and Materials Properties (Springer Science & Business Media, Heidelberg, 2010).
- [282] S. G. Louie, Y.-H. Chan, F. H. da Jornada, Z. Li, and D. Y. Qiu, Discovering and understanding materials through computation, Nat. Mater. 20, 728 (2021).
- [283] K. L. Seyler, D. Zhong, D. R. Klein, S. Gao, X. Zhang, B. Huang, E. Navarro-Moratalla, L. Yang, D. H. Cobden, M. A. McGuire, W. Yao, D. Xiao, P. Jarillo-Herrero, and X. Xu, *Ligand-field helical luminescence in a 2D ferromagnetic insulator*, Nat. Phys. 14, 277 (2018).
- [284] F. Dirnberger, R. Bushati, B. Datta, A. Kumar, A. H. MacDonald, E. Baldini, and V. M. Menon, *Spin-correlated exciton–polaritons in a van der Waals magnet*, Nat. Nanotechnol. **17**, 1060 (2022).
- [285] C. Gong and X. Zhang, *Two-dimensional magnetic crystals and emergent heterostructure devices*, Science **363**, eaav4450 (2019).
- [286] W. He, Y. Shen, K. Wohlfeld, J. Sears, J. Li, J. Pelliciari, M. Walicki, S. Johnston, E. Baldini, V. Bisogni, M. Mitrano, and M. P. M. Dean, *Magnetically propagating*

*Hund's exciton in van der Waals antiferromagnet* NiPS<sub>3</sub>, Nat. Commun. **15**, 3496 (2024).

- [287] C. A. Occhialini, Y. Tseng, H. Elnaggar, Q. Song, M. Blei, S. A. Tongay, V. Bisogni, F. M. F. de Groot, J. Pelliciari, and R. Comin, *Nature of excitons and their ligandmediated delocalization in nickel dihalide charge-transfer insulators*, Phys. Rev. X 14, 031007 (2024).
- [288] K. L. Seyler, D. Zhong, B. Huang, X. Linpeng, N. P. Wilson, T. Taniguchi, K. Watanabe, W. Yao, D. Xiao, M. A. McGuire, K.-M. C. Fu, and X. Xu, Valley manipulation by optically tuning the magnetic proximity effect in WSe<sub>2</sub>/CrI<sub>3</sub> heterostructures, Nano Lett. 18, 3823 (2018).
- [289] D. Zhong, K. L. Seyler, X. Linpeng, R. Cheng, N. Sivadas, B. Huang, E. Schmidgall, T. Taniguchi, K. Watanabe, M. A. McGuire, W. Yao, D. Xiao, K.-M. C. Fu, and X. Xu, Van der Waals engineering of ferromagnetic semiconductor heterostructures for spin and valleytronics, Sci. Adv. 3, e1603113 (2017).
- [290] T. Wang, D. Zhang, S. Yang, Z. Lin, Q. Chen, J. Yang, Q. Gong, Z. Chen, Y. Ye, and W. Liu, *Magnetically-dressed CrSBr exciton-polaritons in ultrastrong coupling regime*, Nat. Commun. 14, 5966 (2023).
- [291] F. Dirnberger, J. Quan, R. Bushati, G. M. Diederich, M. Florian, J. Klein, K. Mosina, Z. Sofer, X. Xu, A. Kamra, F. J. García-Vidal, A. Alù, and V. M. Menon, *Magneto-optics in a van der Waals magnet tuned by self-hybridized polaritons*, Nature (London) **620**, 533 (2023).
- [292] D. N. Basov, M. M. Fogler, and F. J. G. de Abajo, *Polaritons in van der Waals materials*, Science **354**, aag1992 (2016).
- [293] C.-C. Chen, B. Moritz, F. Vernay, J. N. Hancock, S. Johnston, C. J. Jia, G. Chabot-Couture, M. Greven, I. Elfimov, G. A. Sawatzky, and T. P. Devereaux, *Unraveling the nature of charge excitations in* La<sub>2</sub>CuO<sub>4</sub> *with momentum-resolved Cu K-edge resonant inelastic x-ray scattering*, Phys. Rev. Lett. **105**, 177401 (2010).
- [294] M. Moretti Sala, V. Bisogni, C. Aruta, G. Balestrino, H. Berger, N. B. Brookes, G. M. d. Luca, D. Di Castro, M. Grioni, M. Guarise, P. G. Medaglia, F. Miletto Granozio, M. Minola, P. Perna, M. Radovic, M. Salluzzo, T. Schmitt, K. J. Zhou, L. Braicovich, and G. Ghiringhelli, *Energy and symmetry of dd excitations in undoped layered cuprates measured by* Cu L<sub>3</sub> resonant inelastic x-ray scattering, New J. Phys. **13**, 043026 (2011).
- [295] K. Okada and A. Kotani, Copper-related information from the oxygen 1s resonant x-ray emission in low-dimensional cuprates, Phys. Rev. B 65, 144530 (2002).
- [296] K. Okada and A. Kotani, *Theory of oxygen 1s resonant x-ray emission in edge-share-type quasi-one-dimensional cuprates*, J. Phys. Soc. Jpn. **72**, 797 (2003).
- [297] K. Tsutsui, T. Tohyama, and S. Maekawa, *Mott gap excitations and resonant inelastic x-ray scattering in doped cuprates*, Phys. Rev. Lett. **91**, 117001 (2003).
- [298] K. Okada, Effects of Hund's coupling on oxygen 1s x-ray absorption and resonant x-ray emission in onedimensional nickelates, J. Phys. Soc. Jpn. 73, 1681 (2004).
- [299] F. Vernay, B. Moritz, I. S. Elfimov, J. Geck, D. Hawthorn, T. P. Devereaux, and G. A. Sawatzky, Cu *K-edge resonant*

*inelastic x-ray scattering in edge-sharing cuprates*, Phys. Rev. B **77**, 104519 (2008).

- [300] Quanty, https://www.quanty.org/, accessed Jan. 22, 2024.
- [301] Y. Wang, G. Fabbris, M. Dean, and G. Kotliar, *EDRIXS:* An open source toolkit for simulating spectra of resonant inelastic x-ray scattering, Comput. Phys. Commun. 243, 151 (2019).
- [302] EDRIXS, https://github.com/NSLS-II/edrixs, accessed Sept. 9, 2021.
- [303] S. Johnston, E. Khatami, and R. Scalettar, A perspective on machine learning and data science for strongly correlated electron problems, Carbon Trends 9, 100231 (2022).
- [304] Z. Chen, N. Andrejevic, N. C. Drucker, T. Nguyen, R. P. Xian, T. Smidt, Y. Wang, R. Ernstorfer, D. A. Tennant, M. Chan, and M. Li, *Machine learning on neutron and x-ray scattering and spectroscopies*, Chem. Phys. Rev. 2, 031301 (2021).
- [305] A. M. Samarakoon and D. A. Tennant, *Machine learning for magnetic phase diagrams and inverse scattering problems*, J. Phys. Condens. Matter 34, 044002 (2021).
- [306] A. Hariki, M. Winder, T. Uozumi, and J. Kuneš, LDA + DMFT approach to resonant inelastic x-ray scattering in correlated materials, Phys. Rev. B 101, 115130 (2020).
- [307] K. Ishii, K. Tsutsui, Y. Endoh, T. Tohyama, S. Maekawa, M. Hoesch, K. Kuzushita, M. Tsubota, T. Inami, J. Mizuki, Y. Murakami, and K. Yamada, *Momentum dependence of charge excitations in the electron-doped superconductor* Nd<sub>1.85</sub>Ce<sub>0.15</sub>CuO<sub>4</sub>: A resonant inelastic x-ray scattering study, Phys. Rev. Lett. **94**, 207003 (2005).
- [308] S. Kourtis, J. van den Brink, and M. Daghofer, Exact diagonalization results for resonant inelastic x-ray scattering spectra of one-dimensional Mott insulators, Phys. Rev. B 85, 064423 (2012).
- [309] K. Wohlfeld, S. Nishimoto, M. W. Haverkort, and J. van den Brink, *Microscopic origin of spin-orbital separation in* Sr<sub>2</sub>CuO<sub>3</sub>, Phys. Rev. B 88, 195138 (2013).
- [310] U. Kumar, A. Nocera, E. Dagotto, and S. Johnston, Theoretical study of the spin and charge dynamics of two-leg ladders as probed by resonant inelastic x-ray scattering, Phys. Rev. B 99, 205130 (2019).
- [311] S. Johnston, C. Monney, V. Bisogni, K.-J. Zhou, R. Kraus, G. Behr, V. N. Strocov, J. Málek, S.-L. Drechsler, J. Geck, T. Schmitt, and J. van den Brink, *Electron-lattice interactions strongly renormalize the charge-transfer energy in the spin-chain cuprate* Li<sub>2</sub>CuO<sub>2</sub>, Nat. Commun. 7, 10563 (2016).
- [312] A. Geondzhian and K. Gilmore, Generalization of the Franck-Condon model for phonon excitations by resonant inelastic x-ray scattering, Phys. Rev. B 101, 214307 (2020).
- [313] A. Nocera, U. Kumar, N. Kaushal, G. Alvarez, E. Dagotto, and S. Johnston, *Computing resonant inelastic x-ray* scattering spectra using the density matrix renormalization group method, Sci. Rep. 8, 1 (2018).
- [314] K. Zawadzki, A. Nocera, and A. E. Feiguin, A timedependent scattering approach to core-level spectroscopies, arXiv:arXiv:2002.04142.
- [315] S. Jiang, D. J. Scalapino, and S. R. White, *Ground-state phase diagram of the t-t'-J model*, Proc. Natl. Acad. Sci. U.S.A. 118, e2109978118 (2021).

- [316] A. Potapova, I. Pilé, T.-C. Yi, R. Mondaini, and E. Burovski, *Dimensional crossover on multileg attractive-U Hubbard ladders*, Phys. Rev. A 107, 053301 (2023).
- [317] Y. Chen, Y. Wang, C. Jia, B. Moritz, A. M. Shvaika, J. K. Freericks, and T. P. Devereaux, *Theory for time-resolved resonant inelastic x-ray scattering*, Phys. Rev. B 99, 104306 (2019).
- [318] C.-Y. Lai and J.-X. Zhu, Ultrafast x-ray absorption spectroscopy of strongly correlated systems: Core hole effect, Phys. Rev. Lett. **122**, 207401 (2019).
- [319] P. Werner, S. Johnston, and M. Eckstein, Nonequilibrium-DMFT based RIXS investigation of the two-orbital Hubbard model, Europhys. Lett. 133, 57005 (2021).
- [320] J. van den Brink and M. van Veenendaal, *Correlation functions measured by indirect resonant inelastic x-ray scattering*, Europhys. Lett. **73**, 121 (2005).
- [321] L. J. P. Ament, F. Forte, and J. van den Brink, Ultrashort lifetime expansion for indirect resonant inelastic x-ray scattering, Phys. Rev. B 75, 115118 (2007).
- [322] V. Bisogni, M. Moretti Sala, A. Bendounan, N. B. Brookes, G. Ghiringhelli, and L. Braicovich, *Bimagnon studies in cuprates with resonant inelastic x-ray scattering at the O K edge. II. Doping effect in* La<sub>2-x</sub>Sr<sub>x</sub>CuO<sub>4</sub>, Phys. Rev. B **85**, 214528 (2012).
- [323] C. Jia, E. Nowadnick, K. Wohlfeld, Y. Kung, C.-C. Chen, S. Johnston, T. Tohyama, B. Moritz, and T. Devereaux, *Persistent spin excitations in doped antiferromagnets revealed by resonant inelastic light scattering*, Nat. Commun. 5, 1 (2014).
- [324] Y. Lu and M. W. Haverkort, Nonperturbative series expansion of Green's functions: The anatomy of resonant inelastic x-ray scattering in the doped Hubbard model, Phys. Rev. Lett. 119, 256401 (2017).
- [325] T. P. Devereaux, A. M. Shvaika, K. Wu, K. Wohlfeld, C. J. Jia, Y. Wang, B. Moritz, L. Chaix, W.-S. Lee, Z.-X. Shen, G. Ghiringhelli, and L. Braicovich, *Directly characterizing* the relative strength and momentum dependence of electron-phonon coupling using resonant inelastic x-ray scattering, Phys. Rev. X 6, 041019 (2016).
- [326] A. M. Tsvelik, R. M. Konik, N. V. Prokof'ev, and I. S. Tupitsyn, *Resonant inelastic x-ray scattering in metals: A diagrammatic approach*, Phys. Rev. Res. 1, 033093 (2019).
- [327] Y. Matsubayashi and S. Ishihara, Numerical study of resonant inelastic x-ray scattering at the oxygen K edge in insulating cuprates, Phys. Rev. B 107, 075121 (2023).
- [328] G. Ghiringhelli, M. Matsubara, C. Dallera, F. Fracassi, R. Gusmeroli, A. Piazzalunga, A. Tagliaferri, N. B. Brookes, A. Kotani, and L. Braicovich, NiO as a test case for high resolution resonant inelastic soft x-ray scattering, J. Phys. Condens. Matter 17, 5397 (2005).
- [329] M. Winder, A. Hariki, and J. Kuneš, X-ray spectroscopy of the rare-earth nickelate LuNiO<sub>3</sub>: LDA + DMFT study, Phys. Rev. B 102, 085155 (2020).
- [330] K. Higashi, M. Winder, J. Kuneš, and A. Hariki, *Core-level x-ray spectroscopy of infinite-layer nickelate:* LDA + DMFT *study*, Phys. Rev. X **11**, 041009 (2021).
- [331] K. Gilmore, J. Pelliciari, Y. Huang, J. J. Kas, M. Dantz, V. N. Strocov, S. Kasahara, Y. Matsuda, T. Das,

T. Shibauchi, and T. Schmitt, *Description of resonant inelastic x-ray scattering in correlated metals*, Phys. Rev. X **11**, 031013 (2021).

- [332] J. Sun, A. Ruzsinszky, and J. P. Perdew, *Strongly con*strained and appropriately normed semilocal density functional, Phys. Rev. Lett. **115**, 036402 (2015).
- [333] J. Sun, R. C. Remsing, Y. Zhang, Z. Sun, A. Ruzsinszky, H. Peng, Z. Yang, A. Paul, U. Waghmare, X. Wu, M. L. Klein, and J. P. Perdew, Accurate first-principles structures and energies of diversely bonded systems from an efficient density functional, Nat. Chem. 8, 831 (2016).
- [334] H. Yavaş, M. van Veenendaal, J. van den Brink, L. J. P. Ament, A. Alatas, B. M. Leu, M.-O. Apostu, N. Wizent, G. Behr, W. Sturhahn, H. Sinn, and E. E. Alp, *Observation of phonons with resonant inelastic x-ray scattering*, J. Phys. Condens. Matter 22, 485601 (2010).
- [335] S. Moser, S. Fatale, P. Krüger, H. Berger, P. Bugnon, A. Magrez, H. Niwa, J. Miyawaki, Y. Harada, and M. Grioni, *Electron-phonon coupling in the bulk of anatase* TiO<sub>2</sub> measured by resonant inelastic x-ray spectroscopy, Phys. Rev. Lett. **115**, 096404 (2015).
- [336] S. Fatale, S. Moser, J. Miyawaki, Y. Harada, and M. Grioni, *Hybridization and electron-phonon coupling in ferroelectric* BaTiO<sub>3</sub> probed by resonant inelastic x-ray scattering, Phys. Rev. B 94, 195131 (2016).
- [337] L. Braicovich, M. Rossi, R. Fumagalli, Y. Peng, Y. Wang, R. Arpaia, D. Betto, G. M. De Luca, D. Di Castro, K. Kummer, M. Moretti Sala, M. Pagetti, G. Balestrino, N. B. Brookes, M. Salluzzo, S. Johnston, J. van den Brink, and G. Ghiringhelli, *Determining the electronphonon coupling in superconducting cuprates by resonant inelastic x-ray scattering: Methods and results on* Nd<sub>1+x</sub>Ba<sub>2-x</sub>Cu<sub>3</sub>O<sub>7- $\delta$ </sub>, Phys. Rev. Res. **2**, 023231 (2020).
- [338] C. D. Dashwood, A. Geondzhian, J. G. Vale, A. C. Pakpour-Tabrizi, C. A. Howard, Q. Faure, L. S. I. Veiga, D. Meyers, S. G. Chiuzbăian, A. Nicolaou, N. Jaouen, R. B. Jackman, A. Nag, M. García-Fernández, K.-J. Zhou, A. C. Walters, K. Gilmore, D. F. McMorrow, and M. P. M. Dean, *Probing electron-phonon interactions away from the Fermi level with resonant inelastic x-ray scattering*, Phys. Rev. X **11**, 041052 (2021).
- [339] J. N. Hancock, G. Chabot-Couture, and M. Greven, Lattice coupling and Franck-Condon effects in K-edge resonant inelastic x-ray scattering, New J. Phys. 12, 033001 (2010).
- [340] J. J. Lee, B. Moritz, W. S. Lee, M. Yi, C. J. Jia, A. P. Sorini, K. Kudo, Y. Koike, K. J. Zhou, C. Monney, V. Strocov, L. Patthey, T. Schmitt, T. P. Devereaux, and Z. X. Shen, *Charge-orbital-lattice coupling effects in the dd excitation* profile of one-dimensional cuprates, Phys. Rev. B 89, 041104(R) (2014).
- [341] K. Bieniasz, S. Johnston, and M. Berciu, *Beyond the* single-site approximation modeling of electron-phonon coupling effects on resonant inelastic x-ray scattering spectra, SciPost Phys. **11**, 062 (2021).
- [342] K. Bieniasz, S. Johnston, and M. Berciu, *Theory of dispersive optical phonons in resonant inelastic x-ray scattering experiments*, Phys. Rev. B 105, L180302 (2022).

- [343] A. Geondzhian and K. Gilmore, *Demonstration of reso*nant inelastic x-ray scattering as a probe of excitonphonon coupling, Phys. Rev. B **98**, 214305 (2018).
- [344] K. Gilmore, *Quantifying vibronic coupling with resonant inelastic x-ray scattering*, Phys. Chem. Chem. Phys. 25, 217 (2023).
- [345] Y. Wang, M. Claassen, C. D. Pemmaraju, C. Jia, B. Moritz, and T. P. Devereaux, *Theoretical understanding of photon* spectroscopies in correlated materials in and out of equilibrium, Nat. Rev. Mater. 3, 312 (2018).
- [346] Y. Wang, Y. Chen, C. Jia, B. Moritz, and T. P. Devereaux, *Time-resolved resonant inelastic x-ray scattering in a pumped Mott insulator*, Phys. Rev. B 101, 165126 (2020).
- [347] P. Werner and M. Eckstein, *Nonequilibrium resonant inelastic x-ray scattering study of an electron-phonon model*, Phys. Rev. B **104**, 085155 (2021).
- [348] M. Eckstein and P. Werner, *Simulation of time-dependent* resonant inelastic x-ray scattering using nonequilibrium dynamical mean-field theory, Phys. Rev. B **103**, 115136 (2021).
- [349] N. Tancogne-Dejean, M. A. Sentef, and A. Rubio, *Ultra-fast modification of Hubbard U in a strongly correlated material: Ab initio high-harmonic generation in* NiO, Phys. Rev. Lett. **121**, 097402 (2018).

- [350] J. G. Checkelsky, B. A. Bernevig, P. Coleman *et al.*, *Flat bands, strange metals and the Kondo effect*, Nat. Rev. Mater. 9, 509 (2024).
- [351] J. Kim, D. Casa, A. Said, R. Krakora, B. J. Kim, E. Kasman, X. Huang, and T. Gog, *Quartz-based flat-crystal resonant inelastic x-ray scattering spectrometer with sub-10 meV energy resolution*, Sci. Rep. 8, 1 (2018).
- [352] R. Margraf, R. Robles, A. Halavanau, J. Kryzywinski, K. Li, J. MacArthur, T. Osaka, A. Sakdinawat, T. Sato, Y. Sun, K. Tamasaku, Z. Huang, G. Marcus, and D. Zhu, *Low-loss* stable storage of 1.2 Å x-ray pulses in a 14 m Bragg cavity, Nat. Photonics 17, 878 (2023).
- [353] Y. Shvyd'ko, Diffraction gratings with two-ordersof-magnitude-enhanced dispersion rates for sub-meV resolution soft x-ray spectroscopy, J. Synchrotron Radiat. 27, 1227 (2020).
- [354] R. K. Heilmann, A. R. Bruccoleri, V. Burwitz, C. DeRoo, A. Garner, H. M. Günther, E. M. Gullikson, G. Hartner, E. Hertz, A. Langmeier, T. Müller, S. Rukdee, T. Schmidt, R. K. Smith, and M. L. Schattenburg, *X-ray performance of critical-angle transmission grating prototypes for the arcus mission*, Astrophys. J. 934, 171 (2022).